

Temi di discussione

(Working Papers)

Foreign investors and target firms' financial structure: cavalry or lucusts?

by Lorenzo Bencivelli and Beniamino Pisicoli







Temi di discussione

(Working Papers)

Foreign investors and target firms' financial structure: cavalry or lucusts?

by Lorenzo Bencivelli and Beniamino Pisicoli

Number 1327 - April 2021

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Federico Cingano, Marianna Riggi, Monica Andini, Audinga Baltrunaite, Marco Bottone, Davide Delle Monache, Sara Formai, Francesco Franceschi, Adriana Grasso, Salvatore Lo Bello, Juho Taneli Makinen, Luca Metelli, Marco Savegnago. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

FOREIGN INVESTORS AND TARGET FIRMS' FINANCIAL STRUCTURE: CAVALRY OR LUCUSTS?

by Lorenzo Bencivelli* and Beniamino Pisicoli**

Abstract

We study how FDI affects the financial structure of targeted firms, by looking at a sample of foreign acquisitions that occurred in Italy between 1998 and 2016. We show that the entry of foreign investors promotes the diversification of financing sources. Moreover, foreign acquisitions lower investment sensitivity to the availability of bank credit and the cash flow sensitivity of cash, allowing targeted firms to rely more on non-bank external financing channels. Importantly, these effects are stronger for investment in intangible assets. These findings suggest that the positive productivity effects of FDI emphasized in the literature are, at least in part, traceable to enhanced investment in capital that is harder to finance through the banking sector.

JEL Classification: F15, F21, F23, F61.

Keywords: FDIs, firms' financial structure, non-bank financing, investment. **DOI:** 10.32057/0.TD.2021.1327

Contents

1. Introduction	5
2. Literature review	7
3. Data and descriptive statistics	8
4. Econometric strategy	13
4.1 The FDI impact on the financial structure	15
4.2 The FDI impact on investment	17
5. Main findings	20
5.1 Propensity score results	20
5.2 Regression results – foreign vs domestic acquisition	21
5.3 Sensitivity of investments to foreign acquisition	24
6. Some considerations on the transmission channels	29
7. Additional results and robustness analysis	34
7.1 FDIs, financial structure and borrowing costs	34
7.2 More on the target firm's financial constraint	36
7.3 What if the investor is domestic?	38
8. Concluding remarks	40
References	40

^{*}Bank of Italy, Directorate General for Economics, Statistics and Research.

^{**}University of Rome Tor Vergata.

1 Introduction¹

Foreign Direct Investments (FDIs) represent a relevant factor in the globalized economy. According to the UNCTAD, in 2017 the global stock of FDI was more than USD 31.5 trillion, or about 39% of global GDP, 8 percentage points more than 5 years before. Among advanced economies, the stock of inward FDI is 44% of GDP but the picture is rather heterogeneous, with the US performing broadly in line with the average (40% of GDP) and the major European countries well below: France, Italy and Germany with stocks of, respectively, 34, 27 and 25% of GDP.

Through FDIs, investing firms access distant foreign markets, leverage effectively the diffusion of the global value chains, and foster productivity gains. For the recipient firms, FDIs represent not only a channel for the acquisition of technology and managerial practices, but also an important financing source. In the European Union, the free movement of capital is enshrined in the treaty regulating the functioning of the single market, thus making it one of the most open spaces to FDIs. Institutions and members countries acknowledge the relevance of FDIs in enhancing overall competitiveness and productivity, favoring firms' internationalization, accessing new markets, and sustaining employment.

In this paper, we analyze how a foreign acquisition affects the target firms' financing structure and how this reflects in its investment plans. There are several good reasons why this can happen. The first and most obvious is the availability of fresh funds, which the receiving firms can use to finance new investment or to complete pre-existing projects. Foreign investors could also provide the recipient firm with more and better assets to be used as collateral in bank financing operation. The foreign acquisition may be followed by the appointment of a new management team eager to diversify the firm's financial structure. All these measure have the potential to loosen the target firm's financing constraint and, as a result, reducing investment's sensitivity to bank debt or to cash holdings. Our goal is to investigate the shape of this process, to see if and how target firms may benefit from it and how this compares with a domestic acquisition.

For the purpose of this study, we will define FDI as a cross border acquisition, namely an operation where a foreign subject buys a relevant stake in an existing group, subsidiary or branch not previously invested in from abroad.² To properly account for the non-random selection of the target firms, we will rely on a statistical procedure designed to address the selection bias problem, called propensity score matching (PSM henceforth) approach (Rosenbaum and Rubin, 1983). Our sample includes acquisitions involving a target based in Italy

¹The authors are grateful to Alessandro Borin, Pietro Catte, Paolo Conteduca, Francesca Lotti, Federico Cingano, Michele Mancini, Pasquale Scaramozzino, Giovanni Veronese and the participants to the Workshop of the European Network for Research on Investment, Luxembourg November 2019. All remaining errors are the sole responsibility of the authors.

²In the literature, these investments are often called brownfield, as opposed to greenfield, which is when the investor sets up a new entity. The reason to exclude greenfield investments from our study is twofold: (i) the predominance of acquisitions in the total FDI flows, and (ii) most importantly, the need to identify a clear break in the life of the firms in our sample.

over the period 1997 - 2016.

We find that, in the years following the foreign acquisition, acquired firms experience a rotation in their financing structure toward non-bank sources, thus increasing the diversification of their financing structure. In a context characterized by thin capital markets (about 70% of Italy's total financial assets are held by banks) and firms' heavy reliance on bank credit, this result seems particularly relevant also from a policy standpoint. According to Bugamelli et al. (2018), "the underdevelopment of the non-banking segment has limited and is limiting the growth enhancing contribution that could come from startups and innovation more in general". Bank credit is less suitable to finance riskier activities characterized by asymmetric information. In addition, over-reliance on bank borrowing increases investment's sensitivity to exogenous bank supply shocks (Amiti and Weinstein, 2018). Small bank-dependent firms are found to reduce inventory demand, investment, and employment in times of tight monetary policy more than firms able to diversify their debt composition (Kashyap et al., 1994; Gertler and Gilchrist, 1994; Kashyap and Stein, 2000). Following the global financial crisis, the prominence of bank credit as an external financing source has played a key role in curbing investment as a result of the financial shock that hit the Italian banking sector (see Cingano et al., 2016), a pattern in line with the experiences of Japan (Kang and Stulz, 1997) and the US (Slovin et al, 1993) during past crisis episodes. In general, excessively bank-based financial systems are believed to show more exacerbated firms' hold-up, entry deterrence via lobbying, and credit supply volatility problems (ESRB, 2014). Finally, Langfield and Pagano (2016) show that bank-centered financial sectors are highly pro-cyclical, overleveraging the economic system during asset price booms and overtightening during asset price drops, increasing the economy's exposure to a large systemic risk.³ However, as capital markets in Italy are relatively modest and corporate bond issuance is rather limited, we expect that a relevant role in non-bank financing is played by components including leasing and factoring, debt toward the parent company and other instruments.

We investigate the extent to which a foreign acquisition fosters target firms' investment activity as compared to not-acquired firms with similar characteristics. We scrutinize this matter from a financing channel point of view. On aggregate, we find that the FDI has a positive impact on total investment by magnifying the positive effect of non-bank financing on accumulation. Looking separately at investment in tangible and intangible assets, we find diversified effects; tangibles benefit from the occurrence of a foreign acquisition as it reduces the dependence of the investment on internal sources, showing that the financing constraint is less binding than before the FDI. On the contrary, FDIs stimulate intangible assets investment because they induce a greater exposure toward non-bank finance.

The remainder of the paper is structured as follows: in the next section we

³The launch of a Capital Markets Union is meant to enhance the contribution of alternatives to bank lending (equity and bond markets, securitization, lending from insurance companies and asset managers, venture capital and crowdfunding) within a single large European capital market (Veron and Wolff, 2016).

briefly review the literature on FDIs and the impact of the latter on firms' financial structure; Section 3 describes our data and variables of interest; Section 4 addresses the selection bias problem and presents our econometric strategy; Section 5 provides our main results on target firms' financial structure and scrutinize the impact of foreign acquisitions on their investment process; robustness checks and additional results are reported in Section 6. Section 7 concludes.

2 Literature review

The literature analyzing the impact of FDIs on the origin and destination economies is ample and spans across several dimensions. The traditional line of research has focused on the effect of foreign investment on the target firm's performance. Braguinsky et al. (2015) find evidence that the acquisition improves both the target firm's profitability and its productivity. Similarly, Bircan (2019) documents that production plants experience an increase in productivity upon the acquisition by multinational corporations. Importantly, the positive effects brought about by the foreign investor spill over into the entire sector of destination by increasing competition, lowering prices, and pushing less productive firms to exit the market. This corroborates the findings provided by Javorcik (2004) who shows the existence of "backward vertical spillovers" positively affecting the productivity of the sector targeted by the foreign investor. In other words, "vertical" relationships with downstream multinational affiliates enhances industry's productivity. On the contrary, domestic sectors do not seem to benefit from "horizontal" contacts with foreign firms. As for the Italian context, Bentivogli and Mirenda (2017) find that the performance of domestic firms, measured in terms of sales, return on equity, cash flow and indebtedness, improves after a foreign acquisition.

Foreign investors may also affect the innovation activity carried out by the target firm. Guadalupe et al. (2012) find empirical evidence that FDI recipient firms increase their innovation intensity with respect to not acquired firms. Stiebale (2016) shows that cross-border M&As lead to a higher level of innovation by the merged entity. However, the increase of innovation intensity seems to be linked to the characteristics of the parent company; in particular, R&D activities and technology are relocated toward the more suitable environment for innovation following the acquisition. The latter in turn is usually the country of origin of the acquiring company.

A growing strand of literature is investigating the link between FDIs and finance. Some studies have related the outcome of the foreign M&A on the individual firm to the surrounding financial environment. For instance, Alfaro et al. (2004) find that FDIs need well-developed financial markets in the destination country in order to trigger their positive effects. Desai et al. (2004) argue that internal and local external financing are substitutes for the subsidiaries of multinational firms. Subsidiaries operating in jurisdictions with underdeveloped financial markets and weaker creditors' rights are more prone to borrow from the parent company than tapping local markets.

Foreign investment also impacts the recipient firm's and sector's financial structure. Anwar and Sun (2015) investigate the impact on the capital structure of domestic firms in relation to the presence of other foreign firms in the same industry. They document that in sectors with greater foreign participation, domestic firms exhibit a lower financial debt and increase investments. Erel et al. (2015), though not emphasizing the investor's origin (whether foreign or domestic), find that acquisitions tend to relax the financial constraints on the target firms, confirming the idea that financial synergies between the investor and the target can be a determinant underlying the investment decision. Stiebale and Wößner (2019) work along a similar line, controlling for the selection bias induced by the investment process. They show that upon takeover, acquired firms tend to have a better access to external finance, higher tangible and intangible assets, and lower cash holdings. The reduction in cash holdings is linked to the relaxation of the financing constraint faced by the target firm following the acquisition. Khatami et al. (2015) show that in the presence of a financial constraint, both firms, the acquired and the acquirer, benefit significantly from the acquisition. Through the takeover, indeed, the former gains access to investment opportunities otherwise unavailable to it and enjoys a greater acquisition premium, while the latter benefits from unexploited growth opportunities through the relaxation of the target firm's financial constraint.

3 Data and descriptive statistics

Following UNCTAD, an FDI is defined as "an investment reflecting a lasting interest and control by a foreign direct investor, resident in one economy, in an enterprise resident in another economy (foreign affiliate)". In practical terms, all investments leading to the ownership by the foreign entity of a stake equal or greater than 10% of the target entity lie within this definition. The 10% cutoff is consistent with the level currently assumed by the IMF, OECD, UNCTAD and World Bank to exclude foreign portfolio investments that do not ensure managerial control over the target company. However, we acknowledge that such a definition has its own limits. On the one hand it excluds the possibility that a controlling stake may come with a smaller share, as it is the case of very large companies with extremely fragmented shareholders base. On the other, not all the investments above 10% of the capital lead to a significant influence in the firm's management, as it happens when the ownership is concentrated in the hands of very few shareholders.

FDI micro data pose the problems of dealing with false positives and false negatives. The first occur when a multinational group moves the controlling company to a different jurisdiction while remaining de-facto a domestic company. Following these operations, corporate ownership information will display a change in the nationality of owners, thus suggesting that an FDI may have taken place. On the other hand, a foreign entity may choose to operate its investment in a country by first setting up a special purpose financial vehicle within the same jurisdiction.⁴ In this case, corporate ownership data would not show a foreign investment, while in fact the controlling entity is now a foreign one, thus representing a foreign investment.

Zephyr is a database provided by the Bureau Van Dijk,⁵ collecting a variety of information about every single operation on record and, in particular, it gives the identity of the ultimate investor in the form of Global Ultimate Owner (GUO) of the acquiring company. This information allows us to discriminate between domestic and foreign acquisitions reducing considerably the risk of false positives and false negatives.⁶

To our purpose, we retrieved data related to any non-Italian GUO company acquiring a stake leading to a capital share above 10% of an Italian firm. According to these criteria, we classify about 1600 operations as FDIs within the period 1997 – 2016. From now on, we will refer only to investments fulfilling the 10% requirement as FDIs. The 10% threshold excludes about 90 operations, 20 of which related to firms with total assets greater than e 1 billion for which also a stake smaller than 10% might be significant and could be reflected into a certain degree of management power exerted by the stakeholder.

We complement this information with that available from CERVED, which collects balance sheet data from all limited liability companies in Italy (Calligaris, 2015).⁷ Using ownership information from ORBIS (Bureau Van Dijk), we remove from our dataset all firms which already had a foreign shareholder owning at least 10% of the capital before the start of our estimation sample. Those firms can be considered to have already experienced the break coming from a foreign acquisition, and we would not be able to disentangle the FDI effect from their individual characteristics. Finally, to have more reliable information on employment, we pool these sets of data with those sourced from the INPS (the Italian national pension system) enabling us to match the exact number of employees per month for each firm displayed in both datasets. We exclude from the dataset firms operating in the financial sector, as their financing and investment strategies follow specific patterns whose analysis is out of the scope of this paper.

 $^{^{4}}$ In this case, the financial vehicle would be a newly installed company, green field which lies beyond the scope of this paper.

⁵The database has been extensively used in the FDIs literature, e.g. see Hattari and Rajan (2011), Stiebale (2016), Rasciute and Downward (2017). Bollaert and Delanghe (2015) consider Zephyr as a reliable data source on M&As, in particular because of its advantage in reporting sellers' details and when information on multiple-targets and multiple-acquirers deals is needed, a particularly important issue in the context of cross-border deals. Moreover, the database benefits from its broad coverage of European deals.

⁶This is one of the differences between our work and Bentivogli and Mirenda (2017) which employ Unioncamere database. The latter reports with a greater precision the timeline of the corporate ownership events but does not present detailed information on the foreign acquirer.

⁷CERVED collects information from the official data available at the Italian Registry of Companies and records the official financial statements filed at the Italian Chambers of Commerce. Limited companies (so-called società di capitali) furnish data to the latter entities by law, so that all joint stock, partnership limited by shares companies and limited liability (S.p.a., S.a.p.a. and S.r.l.) Italian companies are reported in the database (Calligaris, 2015). The number of firms per year whose balance sheet is reported in CERVED increases from little less than 300,000 in 1993 to about 750,000 in 2016.



Figure 1: Number of acquisition in Italy in 1997-2016

	Avg. number of		Avg a	cquired	Acquisition with		
	acquisitions		stal	ke (*)	recapitalization $(*)$		
	Foreign	Domestic	Foreign	Foreign Domestic		Domestic	
1998-2002	68	94	73	55	49	54	
2003-2008	76	140	85	71	61	62	
2009-2012	64	116	86	84	65	59	
2013-2016	121	155	87	91	64	60	

(*) Percentage points.

Table 1: Frequency of the acquisition in Italy by origin of the acquiring company.

In our data, foreign and domestic investments share a similar pattern through time though, since 2000, domestic operations have been steadily more numerous than foreign ones (Figure 1). As the number of operations surveyed before is negligible, our estimation sample starts in 1998. In the period immediately before the global financial crisis (until 2008) and in the one immediately after the sovereign debt crisis (since 2013), Italy experienced a sizeable relative increase of foreign acquisitions, while the number of domestic operations is somewhat stable at around 140 per year. Both are affected by the outbreak of the first wave of the double dip crisis but start recovering at different moment: domestic acquisitions just after 2009 while foreign ones only after the fading out of the effects of the second dip (see Table 1). We don't have reliable information on the value of the deals in our sample, but we show that both domestic and foreign investors tend to buy a majority stake of target firms. In about 60% of the operations in our sample, the acquirer has injected new capital by, presumably, acquiring newly issued shares.

The largest part of foreign investments accrues to Italy from the US and other large European countries (Figure 2). As Italy is primarily a manufacturing economy, there is little surprise that foreign investments concentrate in the manufacturing sector, roughly one-half of the total (Figure 3). Breaking down such category, we find that mechanics, electrical and electronics, and chemicals



Figure 2: Country of origin of foreign acquisition

are the manufacturing subsectors which attracts most foreign investors (Figure 4).

We take a snapshot of our sample of the firms targeted by a foreign and a domestic acquisition in 2007 and 2015 (the peaks in foreign acquisitions before and after the crisis period, respectively) one year prior to the acquisition to get some insights on the characteristics of the average recipient firm. In 2007, 95 foreign acquisitions and 125 domestic ones took place, while 2015 witnessed the maximum number of both foreign and domestic deals over our sample (132 and 174 operations, respectively); see Table 2.

	Foreign acquisition				Dor	nestic ac	quisition
		Year 2	2007				
Variable	Obs	Mean	Std. Dev.	()bs	Mean	Std. Dev.
Total assets (\notin thousands)	85	157,668	786,418	1	.15	60,999	192,485
Bank debt on assets	62	23.0	18.8		76	27.1	47.5
Ebitda on assets	21	0.1	0.1		48	0.1	0.1
Unit labor cost	80	0.8	1.2	1	.04	0.7	0.8
		Year 2	2015				
Total assets (\notin thousands)	127	$51,\!603$	123,109	1	.50	82,468	605,701
Bank debt on assets	84	24.6	26.6		87	19.7	17.8
Ebitda on assets	24	0.1	0.2		63	0.1	0.2
Unit labor cost	117	0.6	1.2	1	.29	0.5	0.9

Table 2: Characteristics of the firms targeted by an acquisition in 2007 and 2015

Many of the relevant variables show a fat tailed distribution, with extremely high standard deviations. Compared to domestically acquired and to notacquired firms, in both years FDI recipients tend to be larger (in terms of total assets) and to exhibit a lower level of bank debt. On the other hand, unit labor cost shows a blurrier picture. In addition, firms targeted by a domestic acqui-



Figure 3: Sector of destination of foreign acquisition (authors elaboration on Nace Rev. 2 Sections)



Figure 4: Sectors of destination of acquisitions in Italy (manufacturing subsectors)

sition are systematically larger than those who did not receive an investment (though the difference between the two groups is smaller), and less indebted. As the mean of the total assets of the foreign-acquired firms has considerably decreased over time, it is fair to assume that foreign investors have broadened their interest to include also smaller firms. Overall, the evidence provided above clearly points to the presence of a significant ex ante selection issue. This in turn would bias results if we were to compare directly FDI targets and not-acquired firms. We address such a concern by pruning our sample with a PSM procedure before implementing our estimation strategy (see section 4 for details).

Though the process of financial deepening in Italy has not been particularly swift, since the beginning of our sample Italian firms have slowly but progressively reduced their reliance on the banking sector in favor of other financial institutions (such as factoring and leasing companies), capital markets and, in some cases, a greater use of self-financing. These developments were particularly pronounced among foreign invested firms. To show this, we plot in Figure 5 the ratio of non-bank debt on total financial debt for the domestic controlled firms and for those who have received a significant foreign investment.⁸ For the purpose of this paper, we will consider as financial debt the following items: (i) short and long-term bank debt, (ii) debt securities; (iii) short and long term financial debt to other creditors. The latter may include, among the others, non-bank financial institutions and parent or affiliated companies. In the period 1998–2015, non-bank debt increased from 33 to 41% among domestically controlled companies whereas it went up from 45 to 62% among those foreign invested. This process is uneven across sectors: in the manufacturing sector, firms did not change much their financing structure between 1998 and 2015, whereas in public utilities was more pronounced (Table 3).

Clearly, the non-bank financing ratio is defined in the [0,1] interval. Another relevant feature of this variable is that it accumulates on the boundaries of its domain. About one fifth of the firms in our sample has only bank debt and about the same has no bank debt at all; all the remaining follow a hump shaped distribution. This pattern is recurrent over time and across different firms' categories. In the proceedings of our analysis, we will have to consider this feature as traditional estimators may fail to account for the boundedness of the distribution.

4 Econometric strategy

The first aim of this paper is to examine whether target firms are more prone to tapping non-bank sources in order to meet their financing needs after a foreign acquisition. In particular, fixing at t the year of the investment, we focus on the evolution of financial debt's composition from t + 1 to t + 5. This in turn seems a reasonable time span to base our analysis on because: (i) it is broad enough

 $^{^8 \}rm We$ deem significant any foreign investment taking the share controlled by a foreign entity above 10%.

	Non Bank financing by sector				
	NB fin. (1998)	Var. 2015-1998			
Agriculture	0.47	0.08			
Manufacturing	0.28	0.02			
Utilities	0.40	0.14			
Construction	0.37	0.07			
Commerce	0.32	0.08			
Logistics	0.34	0.06			
Lodging and hospitality	0.40	0.06			
ICT	0.39	0.08			
Financial services	0.47	0.00			
Real estate	0.42	0.07			
Professional services	0.39	0.08			
Education and healthcare	0.38	0.06			
Other	0.42	0.05			

Table 3: Average non-bank financing by sector



Figure 5: Average non-bank financing ratio

to capture the investor's influence on strategic decisions by firms, and (ii) it limits the interference of extreme events in a firm's life (e.g. default, successive M&As).

When testing the above hypothesis, we need to tackle two main econometric issues. First, as discussed in Section 3, results might be severely affected by ex ante selection. Foreign investors might be more oriented toward more productive firms (cherry-picking) or distressed targets to restructure them (lemongrabbing). In either case, disregarding such motivations would lead to biased results. Second, our variable of interest, non-bank financial debt on total financial debt, ranges in the interval [0,1] by definition and it accumulates on the boundaries, thus biasing the estimates of standard errors in most of the traditional techniques.⁹

⁹About one fifth of the firms in our sample satisfies its external financing need only through bank borrowing and a similar number has virtually no banks' funding; the distribution of

4.1 The FDI impact on the financial structure

To address the selection bias, we implement a matching procedure that provides us with the robust counterfactual needed to describe the performance of the foreign-acquired firms had they not been acquired. In the literature, one of the most popular choices is the Propensity Score Matching (PSM henceforth) (Rosenbaum and Rubin, 1983) as matching procedure.¹⁰ This technique aims at summarizing a set of observable variables into a single score via a binary regression through which one can select those observations in the control group that are ex ante similar to the ones in the treated group. In our case, first, we define the treatment as the FDI received by a target firm. Second, for each treated firm, we pin down a set of untreated observations - i.e. firms which have never been a target of a M&A operation (neither foreign nor domestic) - that are sufficiently close to the foreign acquired ones on the basis of the covariates summarized by the PSM. Finally, using the obtained match, we compare the ex post performance of the two groups.

We proceed with a logit model in which the dependent variable $\text{FDI}_{i,t}$ is a dummy that takes value 1 if firm *i* has gone through a foreign acquisition in the year *t*, and 0 otherwise. Our baseline specification is the following in equation (1):

$$FDI_{i,t} = f(\alpha + \beta CONTROLS_{i,t-1} + \epsilon_{i,t}) \tag{1}$$

where f(.) is a logit transformation function, $\epsilon_{i,t}$ is a well-behaved error term and *CONTROLS* consists of a set of firm's lagged characteristics that influence the probability of being acquired from a foreign investor. The latter term includes total assets in logs (*Total asset*_{i,t-1}), and average annual employment (*Total employment*_{i,t-1}) in log to account for firm size and recognition (see Dahlquist and Robertsson, 2001); intangible assets in logs (*Intangible asset*_{i,t-1}) and intangible over total assets (*Intangible asset ratio*_{i,t-1}) since foreign players may purse reverse-internalization strategies (Eun et al., 1996); non-bank over total financial debt (*Non-bank financing*_{i,t-1}), cash holdings as a percentage of assets (*Cash asset ratio*_{i,t-1}), leverage ratio (*Leverage ratio*_{i,t-1}) and short term debts over total assets (*Short term total debt*_{i,t-1}) to control for firms' ex ante financial condition (Alquist et al., 2019); finally the age of the firm (*Age*_{i,t-1}) enters the specification in its quadratic form.¹¹

the remaining firms is still bimodal with a half of them displaying less than 10% non-bank financing. These characteristics vary little across time and sectors.

¹⁰See for instance Barba Navaretti and Castellani (2004) and Borin and Mancini (2016).

¹¹Previous works remark the existence of a U-shaped relation between the probability of being acquired from a foreign investor and firm's age. In particular, younger firms tend to be desirable targets for foreign investors because of their greater efficiency in assimilating international knowledge (Naldi and Davidsson, 2014). However, as the firm ages, the probability of being acquired decreases until thresholds of sufficient maturity and reputation are reached by the target (Luo and Tung, 2007; Huberman, 2001).

A variety of matching algorithms is available. Major differences among them lie in the definition of the similarity neighborhoods for the treated units and in the weighting of the control observations. In general, broadening the neighborhoods implies a reduced variance but a higher bias that results from associating less similar firms, and vice versa (Caliendo and Kopeinig, 2008). All in all, we opt for a nearest-neighbor algorithm with replacement. To each FDI-firm (conditional on data availability), the scheme associates the k closest units in the control group. Since we implement a PSM with replacement, a single observation can be used multiple times as a match. For consistency of results, we perform exact matching by sector of activity,¹² year and target's origin defined as the macro-region of residence.¹³ In other words, we match target firms with wholly domestic (and not-acquired) firms operating in the same sector and comparable to the former on the basis of the set of covariates illustrated above as observed on the year of the acquisition.¹⁴ We ask the matching procedure to provide us with one to one and one to ten matching.

To assess if the foreign acquisition influences the amount of debt, apart from its composition, we compute the Average Treatment Effects 1 to 5 years after the investment took place finding no significant evidence. To test the impact of the acquisition on the firm's financing structure, we turn our sight on the ratio of non-bank financial debt to total financial debt $(NBF_-VS_BANK_{i,t+k})$, 1 to 5 years after the acquisition. We regress our dependent variable on a dummy taking value 1 for those units who received a foreign direct investment and the value of the dependent at the time of the acquisition (to account for the sluggishness of financial strategies). Our variable of interest is defined in the interval [0,1] (and clustered on the boundaries of that interval) thus biasing any inference based on OLS estimator. To correct for this characteristic, we consider a fractional regression approach (Papke and Woolridge, 1996) (equation 2):¹⁵

$NBF_{-}VS_{-}BANK_{i,t+k} = \alpha + \beta_1 FDI_{i,t} + \beta_2 NBF_{-}VS_{-}BANK_{i,t} + \epsilon_{i,t+k}$ (2)

In our context, a greater diversification in the financing structure of the firm after the FDI would reflect in a positive and significant value for β_1 . Foreign

¹²Sectors are defined by NACE-rev 2 sections. For a sake of simplicity some sections scarcely populated were aggregated, in particular: (i) agriculture and mining, (ii) public administration, education and healthcare and (iii) all residual sections.

¹³We aggregated Italian regions in four macro-region representing homogeneous areas of the country from an economic standpoint: (i) north-west, (ii) north-east, (iii) center and (iv) south and islands.

¹⁴We require the algorithm not to match FDI firms with domestic companies that in our period of interest have gone through a domestic acquisition. Moreover, we exclude from the control group firms that received an FDI before 1997. The Zephyr database collects information on cross-border deals from 1997 on. The inclusion of firms that went through an FDI before such date would bias our results as we would not be able to assign them a correct value for the dummy FDI. We retrieve information on the ownership structure of Italian firms from the Orbis database, cross such information with the Zephyr one and exclude firms with a foreign ultimate owner holding at least 10 % of the company's stakes that are not tracked from Zephyr (meaning that the deal occurred before 1997).

¹⁵The idea is to model the $E(y|\mathbf{X})$ as a logistic function: $E(y|\mathbf{X}) = exp(\mathbf{X}\beta)/[1 + exp(\mathbf{X}\beta)]$ and to estimate the relevant parameter using quasi-maximum likelihood estimator.

investors may be more open to capital and private debt markets than their Italian counterparts thanks to the experience gained in their country of origin, when the latter are less bank-oriented. In addition, their ties with the Italian banking sector might be looser due to the increased functional distance between the two entities (Alessandrini et al., 2009) and they might exploit their greater access to global sources (Fatemi, 1988). Moreover, as a result of the improved creditworthiness and reputation brought about by the foreign deal (Hardin and Holmes, 2002), target firms might be able to overcome some of the market incentives pushing firms to an excessive reliance on the banking system.¹⁶ Finally, the preference for intra-company sources in MNEs financing (Desai et al., 2004) may be another channel to explain the increase of non-bank financial debts. As any effect on our dependent variable may be merely due to a change in bank debt, the denominator of the ratio, and not to an increase of non-bank financing, we replicate the above specification by replacing $NBF_{-}VS_{-}BANK_{i,t+k}$ with the amount of non-bank financial debt over total assets $(NBF_{i,t+k})$ as dependent variable. ¹⁷

4.2 The FDI impact on investment

Going further, we study how investment decisions are affected by the occurrence of a foreign acquisition and, over time, by the rotation in the financing structure. There are several channels through which an acquisition may boost the investment activity of the target firm. The first is the direct impact: as the new ownership structure reflects in the strategies and in the management of the target firm, this will likely affect also the investment plans, though this may happen in either direction. The same may also affect recourse to non-bank financing (as we discussed in the previous section) through management sophistication or other non-tangible characteristics. Tapping a rather under-exploited source may result in an increased investment capacity, and the change in financing preferences may in turn shift investment towards different directions. Moreover, the new ownership will likely bring new resources (in the form of cash or fresh capital means) thus relieving the target firm's financing constraint

 $^{^{16}}$ For instance, Cantillo and Wright (2000) find that more financially distressed firms opt for bank funding over market sources because they seek for the bank superiority in tight monitoring and reorganizational skills.

¹⁷A second econometric issue that arises when performing a matching procedure relates to the Conditional Independence Assumption (CIA). The PSM alleviates ex ante selection on the observables included as regressors in the binary model but it does not control for unobservable firm characteristics. When comparing financing decisions of the two groups we may capture systematic differences due to unobservables and hence draw mistaken conclusions. As a further robustness exercise, we exploit the panel nature of our dataset and replicate our main specification by combining PSM with a difference-in-differences estimation as a robustness exercise. To this end we implement different specifications. Among the others we estimate a pooled and a fixed effects model. Apart from the treatment dummy, the latter includes year and time dummies and a set of firm-specific controls. In both models we omit $NBF_VS_BANK_{i,t}$ (or $NBF_{i,t}$), i.e. the value of the dependent variable in the year of the acquisition, in order to: (i) avoid the bias of estimators in the frame of dynamic panel models (see Kiviet (1995) for details); and (ii) check consistency of our results and fit of the models to the exclusion of the variable. Results are not reported here for brevity.

and starting new investment plans. Finally, the presence of the new owners will likely be accompanied by new physical assets that can be used to collateralize bank's financing, reducing the sensitiveness of investment to bank debt.

To understand what happens in the aftermath of a foreign acquisition we exploit the panel dimension of our dataset and estimate investment sensitivity to cash flow, bank debt and non-bank financing (Fazzari et al., 1988; Almeida and Campello, 2007; Erel et al., 2015; Stiebale and Wößner, 2019). We also add a term ($Recap_{i,t}$) representing the occurrence of a recapitalization, i.e. a dummy variable taking value 1 in the years when the shareholders' capital has increased by at least 50%. We therefore estimate the model in equation (3) with firms' specific fixed effects and time dummies. The idea underlying the equation is that it embodies all the investment financing channels: capital, internal sources, bank borrowing and non-bank financing.

$$\frac{INVESTMENT}{ASSETS}_{i,t} = \alpha + \beta_1 FDI_after_{i,t} + \beta_2 Recap_{i,t} + \beta_3 (FDI_after_{i,t} * Recap_{i,t}) + \beta_4 Cashflow_{i,t} + \beta_5 (FDI_after_{i,t} * Cashflow_{i,t}) + \beta_6 NBF_{i,t} + \beta_7 (FDI_after_{i,t} * NBF_{i,t}) + \beta_8 \Delta BD_{i,t} + \beta_9 (FDI_after_{i,t} * \Delta BD_{i,t}) + \beta_{10} BD_{i,t-1} + \beta_{11} (FDI_after_{i,t} * BD_{i,t-1}) + \Gamma X_{i,t} + c_i + \epsilon_{i,t}$$
(3)

The dependent variable is the net investment at time t of firm i scaled by the average total assets reported by the firm over the period 1998-2016. Using the average level of total assets responds to the need to reflect the cross-sectional dimension of the panel using a time invariant variable; it also avoids possible feedback effects from investments to total assets. Information on the numerator is retrieved from firms' income statements. $FDI_after_{i,t}$ is a dummy variable that takes value 1 starting from the year t in which the firm i has received a foreign investment, and 0 if it refers to: (i) a control firm (never acquired firm), or (ii) a target firm before the investment takes place.¹⁸ $NBF_{i,t}$ represents the amount of non-bank financing. Bank debt is included as percentage variation over the previous year $(\Delta BD_{i,t})$ and in stock lagged one period $(BD_{i,t-1})$ accounts for the possibility that excessive debt stockpiling may pose a drag on the investment process; finally $\Gamma X_{i,t}$ consists in a set of firm characteristics such as current and lagged total assets (in log) and the annual variation of value added (that controls for future growth opportunities) and c_i is the firm i fixed effects term. We first run specifications in which one interaction at time is included, then we move to the full model. All variables in level are scaled by firm i's

 $^{^{18}}$ Practically speaking, the variable is the interaction between a treatment dummy (1 for FDI firms, 0 for control firms) and a time of the treatment dummy (1 after the treatment, 0 before).

average level of total asset in the sample period.¹⁹

The parameter β_1 captures the direct impact of the acquisition, i.e. the one deriving from the presence of the new entity in the corporate ownership. As we have disentangled the financial determinants of investment, this direct impact has more to do with soft factors (change in strategies, in management, in corporate culture, etc.). The parameters β_4 , β_6 , β_8 and β_{10} are the investments' sensitivities to, respectively, cash flow, non-bank financing, bank debt growth at time t, and bank debt stock at time t - 1. The parameters β_5 , β_7 , β_9 and β_{11} taking statistically significant values and opposite sign with respect to the former would suggest that the occurrence of the acquisition has reduced the sensitivity of investments to these factors.

Several papers point out that foreign investors alleviate the target firm's financial constraint by allowing it to tap internal resources (namely borrowing from the parent company) and increasing the firm's capacity to access external financial markets. This effect is broadly recognized in the frame of emerging economies (Rutkowski, 2006; Chen et al., 2017); Erel et al. (2015) and Stiebale and Wößner (2019) address the issue using a sample of European firms.²⁰ In our model, we try to gauge the extent to which the investment decisions are constrained by financial debt (bank and non-bank) and depend upon cash flow.

Not all investments are alike. Tangible investments, for example, tend to be easier to finance through bank credit because of the presence of a physical asset available as collateral. Conversely, intangible investments, are characterized by a more pronounced uncertainty over the underlying asset's value and duration, and are thus less suitable for bank lending (Dell'Ariccia et al., 2017; Cecchetti and Schoenholtz, 2018). Furthermore, intangible assets serve poorly as collateral because they are difficult to re-sell in case of the firm's default. This in turn increases the cost of bank debt for intangible intensive firms. At the same time, banks may be reluctant to provide credit aimed at investment in intangibles and, considering the recent upward trend of the latter, shift their portfolio allocation away from commercial loans (Dell'Ariccia et al., 2017). Indeed, several studies have found that intangible investment is mainly driven by internal sources of financing (Borisova and Brown, 2013; Sun and Xiaolan, 2019) and equity issuance (Brown et al., 2009). Most of these alternatives require an injection of fresh capital, either to fund directly the new investment or to provide new collateral for new financing.

To address the role of foreign acquisition in financing all types of investment, thus, we re-estimate equation (3) for both tangible (net of divestment) and in-

 $^{^{19}}$ We scale all variables by the firm-specific average total assets instead of current total assets to avoid feedback effects from investment to the independent variables.

 $^{^{20}}$ For instance, focusing on Europe Stiebale and Wößner (2019) find that M&A operations relieve target firm's financial constraint, but results are mainly driven by domestic deals rather than cross-border acquisitions.

tangible investments scaled by firm-specific average total assets. If our intuition is correct, we expect target firms' intangible investments to display a certain degree of sensitivity to the flag variable $Recap_{i,t}$, when interacted with the FDI variable (namely, a positive and statistically significant value for β_3).

The entire econometric procedure is replicated using domestically sourced acquisitions to check if the effects we find are related to the foreign nature of the acquirer. The underlying assumption in this case is that foreign investors bring with them better international capital market access and, therefore, are more able to improve the financing of the target firms.

5 Main findings

5.1 Propensity score results

Through propensity score matching, we prune our population in order to feed the main regressions with a sample of firms displaying similar observable characteristics to those treated. Model covariates are chosen after a data-mining process which attempts to maximize, in addition to the fit of the model, the number of treated observations matched, conditional on the data availability and sample unbalance. The treatment is the occurrence of the FDI, and we implement the procedure matching exactly the target's sector, year of occurrence of the FDI and the target's geographic origin.²¹ All control variables in the model are lagged one period.

Logit regression in PSM is reported in Table (4). The dependent variable is the occurrence of FDI at the moment the investment is carried over, the time period spans from 1998 to 2016.²² All variables are significant at least at the 5% level, except for employment (which is significant at 90% level, though) and age squared, thus excluding non linear effects in the target firm's age.

To check for the reliability of the specification, we have to compare the sample moments before and after the pruning procedure. Ideally, if we properly address the selection bias, the distribution of selected control individuals should be equal to that of the treated individuals. The outcome of the balancing property is reported in Table (5). For all variables, t-tests reject the null hypothesis of equal mean between the treated and the untreated samples at conventional significance level before the matching procedure. However, when we compare the treated and the matched sample, the t-test fails to reject the null hypothesis of equal means for all variable, except for the ratio of non-bank financing, although the respective values get closer.

 $^{^{21}}$ Italy is divided in 4 macro regions: North-west, North-east, Center, South and islands. Four dummy variables were created accordingly.

 $^{^{22}{\}rm The}$ specification includes dummy variables for the year, sector of activity and geographic area of origin of the target company.

	Coef.	Std. Err.	t-test	P-value
Total asset (log)	0.48	0.08	5.97	0.00
Intangible asset (log)	0.17	0.03	6.66	0.00
Intangible asset ratio	1.34	0.32	4.17	0.00
Cash asset ratio	0.38	0.07	5.62	0.00
Total employment (log)	0.06	0.03	1.74	0.08
Age	0.02	0.01	-2.47	0.01
Age squared	0.00	0.00	1.02	0.31
Leverage ratio	0.03	0.01	-5.49	0.00
Short term total debt (on asset)	0.46	0.08	5.72	0.00
Non-bank financing (on total debt)	0.25	0.10	-2.59	0.01
Constant	-14.29	0.49	-29.35	0.00
Nobs	2,641,703			
Pseudo R^2	0.17			
Likelihood ratio	-6536.23			

Table 4: Logit model statistics; dependent variable FDI

	Unb	alanced samp	ole	Bal	Balanced sample		
	Treated	Untreated	t-test	Treated	Untreated	t-test	
TTot. ass.	9.56	7.08	***	9.74	9.73		
Int. ass.	5.92	2.95	***	6.26	6.10		
Int. ass. ratio	0.10	0.06	***	0.09	0.09		
Cash ass. ratio	-5.06	-2.74	***	6.17	6.61		
Tot. emp.	4.00	1.98	***	214	201		
Age	16.39	12.14	***	18.54	18.87		
Age squared	441.39	271.11	***	505	507		
Leverage ratio	5.84	9.25	***	4.90	5.24		
ST tot. debt	8.82	6.45	***	39461	36820		
Non-bank fin.	0.41	0.30	***	0.37	0.29	***	

Table 5: Comparison between the sample moments of the treated and control individuals in the whole sample and the matched sample

5.2 Regression results – foreign vs domestic acquisition

In this section, we analyze the evolution of the financial structure of the target firms in reaction to a foreign acquisition at different time horizons after the investment (1 to 5 years). For our purpose, we take the non-bank financing ratio as the dependent variable in a fractional regression approach, to cope with the fact that the variable is defined between 0 and 1 and clustered on the boundaries of this interval. To check the consistency of our results, we feed the regression equation with the sample pruned with the propensity score matching procedure in two different ways: first, we use a sample composed of "twin pairs" which includes the best match for each treated firm, then we use the ten closest observations for each treated firm. Finally, we apply the entire procedure to domestic acquisitions to check whether results are to a large extent due to the nationality of the investors or to the investment itself. The equation takes the following form:

$$Y_{i,t+k} = \alpha + \beta_1 F D I_{i,t} + \beta_2 Y_{i,t} + \epsilon_{i,t+k} \tag{4}$$

where $Y_{i,t+k}$ represents the value of the ratio of the non-bank financing k period after the acquisition took place, $FDI_{i,t}$ is a dummy variable taking value 1 for those firms who received an FDI. Regression results are reported in Table (6).

Κ	Nobs	Pseudo R^2	FDI	t_0	Margin effect					
	1 to 1 matching									
1	2,111	0.311	0.154^{*}	4.023***	0.021^{*}					
2	1,829	0.242	0.241^{***}	3.447^{***}	0.036^{***}					
3	$1,\!607$	0.169	0.197^{**}	2.767^{***}	0.034^{**}					
4	1,411	0.144	0.230^{**}	2.509^{***}	0.041^{**}					
5	1,242	0.122	0.205^{**}	2.333***	0.038**					
		10	to 1 matchi	ng						
1	$10,\!428$	0.3916	0.181*	4.709***	0.021*					
2	9,176	0.312	0.1443	4.004^{***}	0.0197					
3	8,068	0.257	0.171	3.528^{***}	0.026					
4	7,167	0.221	0.094	3.205^{***}	0.015					
5	$6,\!275$	0.185	-0.041	2.891^{***}	-0.007					

Table 6: Fractional regression results for foreign acquisition

All coefficients related to the treatment variable after the 1-to-1 matching procedure are statistically significant, and so are the marginal effects; the share of explained variance spans between 12 and 31%. The results are less significant in case of the 10-to-10 matching procedure, the less accurate one, probably due to difficulties in identifying higher numbers of comparable firms. According to these results, foreign acquisitions impact significantly on the financial structure of the target firms, increasing the share of non-bank financing on total financial debt. The cumulative effect is increasing over time, though not monotonically, and varies between 2 and 4 percentage points. These results are in favor of the hypothesis that receiving an FDI contributes to a greater diversification in the financing sources, which may include the funding from the parent company.²³ To see if our results depend on the foreign nature of the investor, we run the same procedure using domestic acquisition as treatment variable (*DOMESTIC*).

 $^{^{23}}$ As our variable of interest is the ratio of non-bank financial debt to bank debt, the increase in the variable we document might depend on a mere decrease of bank debt. To rule out such possibility we replicate the above procedure by replacing the dependent variable with the ratio of non-bank financial indebtedness to total assets. Results are confirmed. In addition, as a robustness exercise we adopt a difference in differences approach. In particular, we combine PSM with a panel diff-in-diff and estimate several pooled and fixed effects models. We first scrutinize the impact of a foreign direct investment from t - 1 to t + 1, t + 2 and t + 5. Then we investigate the spans t - 1 to t + 1, t - 2 to t + 2 and t - 5 to t + 5. Our findings are fully corroborated for both NBF_VS_BANK and NBF. Target firms significantly increase their exposure to non-bank financial sources after a foreign acquisition. We do not report results here for brevity.

Κ	Nobs	Pseudo R^2	DOMESTIC	t_0	Margin effect					
	1 to 1 matching									
1	2,099	0.288	0.073	3.820^{***}	0.01					
2	1,821	0.217	0.190^{**}	3.228^{***}	0.030^{**}					
3	1,592	0.158	0.146	2.682^{***}	0.026					
4	$1,\!423$	0.139	0.198^{**}	2.483^{***}	0.036^{**}					
5	$1,\!235$	0.119	0.16	2.305^{***}	0.03					
		1	0 to 1 matching							
1	10,376	0.389	0.106^{*}	4.639^{***}	0.022*					
2	9,081	0.304	0.174	3.913^{***}	0.0241					
3	$7,\!990$	0.245	0.173	3.418^{***}	0.027					
4	7,087	0.205	0.071	3.058^{***}	0.012					
5	$6,\!184$	0.181	-0.059	2.863^{***}	-0.01					

Results are in Table (7), while propensity score matching results are not reported for brevity.

Table 7: Fractional regression results for domestic acquisition

Compared to the case of foreign acquisition the magnitude of the increase in non-bank financing is considerably smaller, and significant only in few cases distributed non monotonically throughout the projection horizon (k = 1...5).

Combining these results together, it appears that foreign investors are able to operate a more profound rotation in the financing structure of the target firms as compared with domestic ones. This might be possibly due to the fact that having the backing of firms operating already at a multinational level with greater managerial complexity, they push Italian companies to source financing via a more diversified range of instruments. Moreover, the acquisition by a foreign investor has two effects on the domestic firm from an informational point of view which go in the same direction though having different impacts on the access to financing. On the one hand, it increases the firms' reputation and creditworthiness, allowing the firm a greater access direct borrowing in the form of public bonds or commercial papers (Diamond, 1991; Hale and Santos, 2008). On the other hand, it reduces the amount of soft information banks are able to extract from the firm, which is crucial in relationship lending by banks, a widespread phenomenon in Italy, pushing the firm away from this financing channel (but not necessarily from bank lending, as long as it is not on a relationship basis). Bolton et al. (2016) document that more than half of the 173,879 Italian bank-firm credit relationship considered in their analysis involve a relationship-bank. Several studies show that the ability of banks to build a stable relationship with firms is a function of the physical distance between the two entities (Hauswald and Marquez, 2006, Agarwal and Hauswald 2010). In general, the closer the bank and firm, the easier is for the bank to gather soft information; this, in turn, leads to the development of stable credit relationships. As after the entry of a foreign investor domestic firms may delegate relevant decision power to the former, banks may have a harder time collecting soft information and hence may be reluctant to accommodate the firm's financing needs. This implies that FDI firms may find it easier to rely on non-bank financial debt than to bank indebtedness. In section 6, we provide tentative evidence that show that both the greater financial sophistication of the investor and the distance hypothesis might be plausible explanations of the increased reliance on non-bank financing by target firms.

The next step is to clarify how the shift in the composition of the firm's debt showed up, especially because throughout the time period spanned by our sample, the Italian corporate bond market was particularly underdeveloped, with an extremely low level of issuance from non-financial corporation. In this context, firms may have switched from bank credit to debt vis-à-vis the parent company or may have broadened the set of financing instruments through leasing, factoring and other non-standard measures. In Figure (6), we represent the evolution of the composition of the firms' financial debt in the 1-to-1 matched sample up to 5 years after the investment. For acquired firms, the dynamics appear to be driven mainly by other non-bank financing means (increasing from 29 to 43% as share of total debt), whereas obligations and debt versus stakeholders appear to cover broadly a constant share of the financial burden. Conversely, firms who did not receive a foreign investment display a less pronounced dynamic in the debt composition.



Figure 6: The evolution of the firms' financial structure after a foreign (upper panels) and domestic acquisition (lower panels)

5.3 Sensitivity of investments to foreign acquisition

In this section we analyze the impact of foreign acquisitions on investment by adopting a financing channel perspective. The occurrence of a new investor in the corporate ownership can impact on the different sources of funding in different ways, for example by providing new resources to collateralize bank borrowing or by loosening a possible liquidity constraint faced by the firm, thus reducing the reliance on internal resources such as cash flow. We therefore run the regression in Equation (3) using total investment as dependent variable. The sample is defined by all the matched firms throughout the time period spanning between 1998 and 2016. Results are reported in Table (8).

	-0.008	-0.003	0.011**	0.016***	0.015***	-0.036***
FDI	(0.007)	(0.000)	(0.004)	(0.004)	(0.005)	(0.008)
	-0.002	0.003	0.002	0.002	0.002	-0.001
Recap	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)
~	0.001***	-0.000	0.001***	0.001***	0.001***	-0.000
Cash flow	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
	0.078***	0.078***	0.054***	0.078***	0.078***	0.051***
Non-bank financing	(0.007)	(0.007)	(0.010)	(0.007)	(0.007)	(0.010)
	8.76e-06***	8.69e-06***	8.68e-06***	3.94e-06	8.84e-06***	4.09e-06
Bank debt (growth)	(2.68e-06)	(2.67e-06)	(2.68e-06)	(7.60e-06)	(2.68e-06)	(7.58e-06)
	-0.009	-0.008	-0.008	-0.008	-0.009	-0.010
Bank debt (on asset, lag)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)	(0.006)
	0.031***	(0.000)	(0.000)	(0.000)	(01000)	0.033***
FDI#Recap	(0.007)					(0.007)
	(0.001)	0.002^{***}				0.002***
FD1#Cash flow		(0.000)				(0.000)
		(0.000)	0.046^{***}			0.049***
FDI#Non-bank fin.			(0.013)			(0.013)
			()	5.57e-06		5.05e-06
FDI#Bank debt growth				(8.12e-06)		(8.10e-06)
				()	0.003	0.005
FDI#Bank debt (on asset,lag)					(0.011)	(0.011)
Observations	20,659	20,659	20,659	20,659	20,659	20,659
Number of firms	1,730	1,730	1,730	1,730	1,730	1,730
Constant	yes	yes	yes	yes	yes	yes
Year dummies	yes	yes	yes	yes	yes	yes
Additional firm-level controls	yes	yes	yes	yes	yes	yes
R^2 within	0.156	0.159	0.155	0.155	0.155	0.161
R^2 overall	0.061	0.058	0.060	0.060	0.060	0.060
F-test	108.9	111.9	108.7	108.3	108.3	100.6
Prob(F-test) > F	0.000	0.000	0.000	0.000	0.000	0.000
NOTED DI LIGGO COLO LL		1 . 1			(

NOTES: Period 1998-2016. Additional firm-level controls: current and lagged total assets (in log), annual variation of value added. Standard errors in parentheses. *** p0.01, ** p0.05, * p0.10

Table 8: Panel regression results. Dependent variable: Total Investments.

Although the coefficient on the direct effect is not always positive, the FDI variable turning 1 (from 0) has a positive marginal impact on total investment, especially when FDI is combined with an increase in shareholder capital. However, we fail to replicate the results by Stiebale and Wößner (2019): total investment is sensitive to cash flow, indicating the possible presence of a financing constraint (though the coefficient is very small in size), but the occurrence of a foreign acquisition appears to amplify such sensitivity. Non-bank financing is also positively related to total investment, and the event of a foreign acquisition increases by 25% the investment sensitiveness to this form of financing. This suggests that the target firms rely even more heavily on alternative sources of funding than they used to do prior to the acquisition, as new resources became available through this channel. Total investment responds positively to the growth in bank debt, but such reliance is no more relevant after the entry of a new investor in the firm's ownership structure. This fact may reflect the shift in investment financing from bank borrowing to alternative sources. Past bank debt dose not represent a significant impediment to investment.

Next, we distinguish between tangible and intangible investment. Tangible investment behavior in response to a foreign acquisition is reported in Table (9).

Recapitalization along with the foreign acquisition remains relevant in boosting target firms' investment activity. Tangible investments are insensitive to cash flow; this result can be attributed to the fact that the presence of a collateral loosens the need to tap internal resources. In addition, the foreign investor increases the already positive relevance of non-bank financing as a source of investment in tangible assets, though understandably this happens to a lower extent compared to immaterial assets. In the case of material investments, forms of financing such as leasing may play a significant role. The occurrence of a foreign investment increases the relevance of this channel. Past bank debt drags on the amount of tangible investments.

EDI	-0.015^{***}	-0.003	-0.008***	-0.004^{*}	-0.001	-0.013**
FDI	(0.005)	(0.003)	(0.003)	(0.003)	(0.003)	(0.005)
	-0.001	0.000	0.000	0.001	0.000	-0.001
Recap	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)
	-7 30e-05	-4 89e-06	-7 97e-05	-7 97e-05	-8 22e-05	-9 70e-06
Cash flow	(8 94e-05)	(0.000110)	(8.94e-05)	(8.94e-05)	(8.94e-05)	(0.000110)
	0.024***	0.024***	0.008	0.024***	0.025***	0.009
Non-bank financing	(0.005)	(0.005)	(0.007)	(0.005)	(0.005)	(0.007)
	-1.66e-06	-1.63e-06	-1.73e-06	-1.220-06	-1.73e-06	-1.220-06
Bank debt (growth)	(1.81e-06)	(1.81e-06)	(1.81e-06)	(5.15e-06)	(1.81e-06)	(5.15e-06)
	-0.018***	-0.018***	-0.018***	-0.018***	-0.015***	-0.015***
Bank debt (on asset, lag)	(0.004)	(0.004)	(0.004)	(0.004)	-0.015	(0.004)
	0.004)	(0.004)	(0.004)	(0.004)	(0.004)	0.004)
FDI#Recap	(0.005)					(0.005)
	(0.005)	0.000				0.003)
FDI#Cash flow		-0.000				-0.000
		(0.000)	0.020***			0.000)
FDI#Non-bank fin.			(0.00000)			(0.029
			(0.00892)	470-07		(0.00893)
FDI#Bank debt growth				-4.70e-07		-7.05e-07
				(0.49e-00)	0.019*	(0.000-00)
FDI#Bank debt (on asset,lag)					-0.013	-0.015
	20.050	00.050	00.050	00.050	(0.007)	(0.007)
Observations	20,659	20,659	20,659	20,659	20,659	20,659
Number of firms	1,730	1,730	1,730	1,730	1,730	1,730
Constant	yes	yes	yes	yes	yes	yes
Year dummies	yes	yes	yes	yes	yes	yes
Additional firm-level controls	yes	yes	yes	yes	yes	yes
R^2 within	0.103	0.102	0.103	0.102	0.102	0.103
R^2 overall	0.039	0.040	0.040	0.039	0.039	0.041
F-test	67.52	67.30	67.65	67.26	67.36	60.49
Prob(F-test) > F	0.000	0.000	0.000	0.000	0.000	0.000

NOTES: Period 1998-2016. Additional firm-level controls: current and lagged total assets (in log), annual variation of value added. Standard errors in parentheses. *** p0.01, ** p0.05, * p0.10

Table 9: Panel regression results. Dependent variable: Tangible investments

Results related to intangible investments are presented in Table (10). Intuitively, intangible asset investments rely predominantly on non bank financing, while the effect of cash flow and bank financing are much smaller, though statistically significant. The occurrence of a foreign acquisition influences the intangible investments mainly through internal resources, namely through the recapitalization and by doubling the relevance of cash flow as financing mean. Combining this result with those the first part of the paper related to the shift in sources of financing suggests that the arrival of a foreign investor is positive for intangible investment, usually harder to finance through bank borrowing, also in a context, like the Italian one, strongly centered on the banking sector for the financing of the real economy.

In Figure (7) we have reported the marginal effects of turning the variable

IDI	0.007	0.000	0.019***	0.020***	0.016***	-0.030***
FDI	(0.005)	(0.003)	(0.003)	(0.003)	(0.004)	(0.006)
Deser	-0.001	0.003	0.001	0.001	0.001	0.001
Recap	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)
G 1 8	8.84e-04***	-1.47e-04	8.76e-04***	8.75e-04 ^{***}	8.79e-04 ^{***}	1.58e-04
Cash flow	(1.01e-04)	(1.23e-04)	(1.00e-04)	(1.00e-04)	(1.01e-04)	(1.23e-04)
New Level Generation	0.053***	0.053^{***}	0.046***	0.054^{***}	0.054***	0.042***
Non-Dank financing	(0.005)	(0.005)	(0.008)	(0.005)	(0.005)	(0.007)
	1.04e-05***	1.03e-05***	1.04e-05***	5.16e-06	1.06e-05***	5.31e-06
Bank debt (growth)	(2.04e-06)	(2.03e-06)	(2.04e-06)	(5.79e-06)	(2.04e-06)	(5.75e-06)
Deal delt (an exact lea)	0.009**	0.010**	0.010**	0.010**	0.006	0.004
Bank debt (on asset, lag)	(0.004)	(0.004)	(0.004)	(0.004)	(0.005)	(0.005)
EDI #D	0.017***					0.019***
FDI#Recap	(0.006)					(0.006)
EDL//Cost 4	. ,	0.003^{***}				0.003***
FDI#Cash now		(0.000)				(0.000)
EDI//New head for			0.016			0.020**
FDI#Non-bank fin.			(0.010)			(0.010)
				6.04e-06		5.75e-06
FDI#Bank debt growth				(6.18e-06)		(6.15e-06)
EDI//Deal date (an exact lea)					0.016*	0.019**
FDI#Bank debt (on asset, lag)					(0.009)	(0.008)
Observations	20,659	20,659	20,659	20,659	20,659	20,659
Number of firms	1,730	1,730	1,730	1,730	1,730	1,730
Constant	yes	yes	yes	yes	yes	yes
Year dummies	yes	yes	yes	yes	yes	yes
Additional firm-level controls	yes	yes	yes	yes	yes	yes
R^2 within	0.080	0.089	0.079	0.079	0.079	0.090
R^2 overall	0.033	0.033	0.032	0.032	0.032	0.034
F-test	67.52	67.30	67.65	67.26	67.36	60.49
Prob(F-test) > F	0.000	0.000	0.000	0.000	0.000	0.000

NOTES: Period 1998-2016. Additional firm-level controls: current and lagged total assets (in log), annual variation of value added. Standard errors in parentheses. *** p0.01, ** p0.05, * p0.10

Table 10: Panel regression results. Dependent variable: Intangible investments

 $FDI_after = 1$ (all other variables are at their respective mean). The foreign acquisition has a positive impact at the margin for total and intangible investments. In Figure (8) we reported the adjusted marginal predictions on investments as a function of non bank financing conditional on the interaction between FDI_after and Recap. All types of investment respond positively to an increase of non bank financing. The occurrence of a foreign acquisition increases the responsiveness of investments to an increase in non-bank financing. The coincidence of a foreign acquisition with a recapitalization magnifies this responsiveness.



Figure 7: Marginal effects of FDI on total (left panel), tangible (middle panel) and intangible investments (rigth panel).



Figure 8: Adjusted predictions. Effect of non-bank financing interacted with FDI and Recap on total (top panel), tangible (middle panel) and intangible investments (bottom panel).

6 Some considerations on the transmission channels

The literature provides a number of reasons why firms shift their preferences progressively away from bank financing. First, foreign investors may be willing to diversify their financing sources drawing on their own experience, either as a multinational firm or as a firm coming from a more sophisticated country in terms of financial development. We refer to such hypothesis as to a *sophistication* hypothesis. Another possibility is that the arrival of a foreign entity increases the physical and the functional distance between the firm and its reference bank. In the context characterized by relationship lending (Bolton et al., 2016) where firm-bank distance matters (Hauswald and Marquez, 2006, Alessandrini et al., 2009, Agarwal and Hauswald 2010), the arrival of the foreign investor loosens the ties between the two entities and this requires to increase the funding from other sources. We call this the *distance* hypothesis.²⁴

To explore these hypotheses, we take a closer look to FDI recipient firms in our panel regressions. To test the *sophistication* hypothesis we regress nonbank financing on a set of variables including: (i) a dummy taking value 1 if the acquirer's country of origin is a developed one,²⁵ (ii) the value of stocks traded in the capital market of the acquirer country as share of GDP.²⁶ Here we assume that investors coming from high income economies are more familiar with a diversified set of financial instruments. In the same vein, stock market depth serves as a proxy for market-oriented economies, opposed to bank-based ones. To the purpose of our analysis, we expect both variables to be positively related to the NBF_VS_BANK showed by the target firm after the investment, i.e. the more financially sophisticated the country of origin of the acquirer, the greater the reliance on non-bank financing of the target firm.

As for the second hypothesis, we interpret distance in three ways: (i) functional (operational), (ii) cultural, and (iii) physical distance. The first has to do with the amount of of capital the investor acquires and how this reflects in the governance power, making the target firm more or less distant from the bank of reference. To this purposes, we will use the size of the share bought by the acquiring firm, call it (*Stake*). To capture cultural distance, that regards the differences in habits, values and the legal system between the Italian context and foreign ones, we will use the binary variable *Common Law* from La Porta et al. (1999). It takes value 1 if the country of origin of the investment adopts a common law legal system, as opposed to the civil law system set up

²⁴Other hypotheses draw on the firms improved capacity to tap capital markets. Fatemi (1988) suggests that FDI recipient firms are able to diversify their financial structure because of the increased access to global financial markets. Moreover, the foreign investor might increase target firms' reputation, hence gaining a greater access to direct borrowing in the form of public bonds or commercial papers (Diamond, 1991; Hale and Santos, 2008).

 $^{^{25}\}mathrm{We}$ employ the list of high-income economies based on 2019 data as defined in the frame of the World Bank Country and Lending Groups. Data retrieved from https://datahelpdesk.worldbank.org/knowledgebase/articles/906519

 $^{^{26}\}mathrm{Data}$ come from the World Bank World Development Indicators. The variable is an average in the period 1998-2016.

in Italy. Finally, we use a number of variables from Mayer and Zignano (2011) to measure physical distance between the country of origin of the investor and Italy. In particular, we retrieve information on the distance between capital cities, main cities, and two indexes of distance between the biggest Italian and foreign city, differently weighted by the share of the city in the overall country's population (see Mayer and Zignago (2011) for further details). We also build a dummy (ExtraEU) that takes value 1 if the investor comes from an extra European Union country, 0 otherwise. This variable in a sense rests in the middle between physical and cultural distance (since European Union countries share to a certain degree values and legal frameworks). Even if the three concepts of distance we employ might be imperfect and prone to interrelations, in general we expect them to be positively related to the usage of non-bank sources. This might relate to the effect that a greater distance between the investor and the banking sector produces on the informational profile of the target firm. With the increased distance, the bank's capacity to collect soft information from the firm deteriorates and hence they become more reluctant to accommodate the firm's financing needs. This might produce the shift of FDI firms towards non-bank financial debt.

Table 11 reports the coefficient estimates of the projection of the non-bank financing (as ratio of total financial debt) on functional and physical proxies of distance, and financial sophistication variables of the country of origin of the investor. We limit our attention to observations related to the year in which the investment takes place. Results are positive and strongly significant for *Stake*. The greater the acquired stake, the more the target relies on non-bank financing, in line with our *distance* hypothesis. The impact of *ExtraEU* is statistically non-significant, indicating that the cultural distance may be less relevant tan the functional distance. Both *Developed* and the orientation towards capital markets of the country of origin of the investor show a positive and stable coefficient, suggesting that the *sophistication* may have some ground in explaining the predominance of alternatives to bank credit as funding source.²⁷

In Table 12 we extend the analysis to our core cultural and physical distance measures. *Stake* and *Developed* present again robust positive coefficients. The *Common Law* dummy has a positive though not statistically significant effect on the non-bank financing. This might be possibly due to interrelations with the developed dummy. Moreover, results seem to suggest that the greater the physical distance between the investing firm and Italy, the more the target firms recurs to non-bank financing, as expected.

Finally, in Table 13 we consider not only observations referred to the year of the investment, but also the following history of the target firm. The effect of *Stake* is always positive and strongly significant, in line with the *distance* hypothesis in the functional sense. However, cultural variables provide a fuzzy picture while the impact of physical distance proxies vanishes, also when we limit observations from the year of the acquisition to five years after. Hence,

 $^{^{27}}$ Results are similar when we impose a different lag structure of NBF_VS_BANK and we define different time windows, e.g. one year after the FDI for the dependent variable and NBF_VS_BANK in the year before the FDI as regressor.

Stake		0.007***	0.006***	0.006***	0.009***
		(0.002)	(0.002)	(0.002)	(0.002)
Developed			0.620^{***}		0.771^{***}
			(0.223)		(0.246)
ExtraEU				-0.100	0.226
				(0.135)	(0.147)
NBF_VS_BANK (lag)	2.675^{***}	2.669^{***}	2.650^{***}	2.666^{***}	2.520^{***}
	(0.169)	(0.183)	(0.183)	(0.183)	(0.207)
Mark.Cap./(%GDP)	0.002^{**}	0.002^{**}	0.002	0.002^{**}	
	(0.001)	(0.001)	(0.001)	(0.001)	
Observations	1,010	894	894	894	718
Constant	yes	yes	yes	yes	yes
Year dummies	yes	yes	yes	yes	yes
Pseudo R2	0.162	0.166	0.169	0.167	0.167
F-test	5.886	20.23	26.90	20.41	24.98
Prob(F-test) > F	0.005	0.000	0.000	0.000	0.000

NOTES: Dependent variable: NBF-VS_BANK. The sample consists of the FDI recipient firms used in the panel regressions. Mark.Cap. stands for the average between 1998 and 2016 of the value of stocks traded in capital markets over GDP in the country of origin of the investor. Standard errors in parantheses. *** p<0.01, ** p<0.05, * p<0.10

Table 11: The impact of distance and financial sophistication on non-bank indebtedness. Fractional regressions. Dependent variable: $NBF_{-}VS_{-}BANK$

only functional distance seems to matter in the longer term. A possible explanation of this result is that when the foreign investor has the time to settle in the new Italian environment, its cultural and physical distance with the latter decreases, so that such considerations do not matter anymore for the ability to build relationships with the banking sector.²⁸ Our findings suggest that also the financial sophistication of the country of origin matters. If the investor comes from a country in which firms do not excessively rely on bank lending, as denoted by a higher trading activity of stocks, the orientation to non-bank financing of the target firm is more marked. At the same time, this is valid also for the *Developed* dummy. Both variables are positive and strongly significant, corroborating our *sophistication* hypothesis.

To sum up, our econometric exercises point to the potential validity of both the *sophistication* and *distance* hypotheses. As for the latter, it seems that it is distance in functional terms that matters the most, while cultural and physical distance play a minor role in determining the target firm's financial structure over time. Reputational effects might be at work too. However, our data do not enable us to investigate such issue, so that we delegate this to future extensions.

 $^{^{28}}$ The exercises on distance rest on two assumptions. Since we are interested on the distance between the *firm* and the Italian banking sector, we need *effective* measures of distance between the firm and Italy. While the acquired stake can be easily interpreted as an effective measure of distance of the firm, we need to assume that measures of distance between the *investor* and Italy will reflect the distance between the *firm* and Italy. This can be assumed to be true in the year of the investment. However, it is not clear whether this holds completely also in the years after the acquisition. The second assumption is that the majority of bank debt reported by the firm comes from Italian institutions. While this might not be the case, with our data we are not able to distinguish between Italian and foreign sources of debt.

NBF_VS_BANK (lag)	2.649***	2.657***	2.657***	2.658^{***}	2.658^{***}
	(0.184)	(0.183)	(0.183)	(0.184)	(0.184)
Stake	0.007***	0.007***	0.007***	0.007***	0.007***
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
Developed	0.683***	0.765^{***}	0.769* ^{***}	0.765^{***}	0.766^{***}
-	(0.217)	(0.221)	(0.222)	(0.221)	(0.221)
Common Law	0.145	, ,	,	,	,
	(0.119)				
Distance (capital)	. ,	$3.19e-05^*$			
× - /		(1.84e-05)			
Distance (main city)		,	$3.25e-05^*$		
· · · · · · · · · · · · · · · · · · ·			(1.88e-05)		
Distance (weighted)			. ,	$2.98e-05^*$	
				(1.67e-05)	
Distance (weighted, Ces)				,	$2.98e-05^*$
					(1.68e-05)
Observations	895	895	895	895	895
Constant	yes	yes	yes	yes	yes
Year dummies	yes	yes	yes	yes	yes
Pseudo R2	0.169	0.170	0.170	0.170	0.170
F-test	27.16	29.01	28.98	29.14	29.13
Prob(F-test) > F	0.000	0.000	0.000	0.000	0.000

Table 12: The impact of distance and financial sophistication on non-bank indebtedness. Fractional regressions. Dependent variable: NBF_VS_BANK

	Full panel	Max 5 years after FDI	Full panel	Max 5 years after FDI	Full panel	Max 5 years after FDI	Full panel	Max 5 years after FDI
$NBF_VS_BANK (lag)$	3.779^{***}	3.441^{***}	3.808^{***}	3.472^{***}	3.776^{***}	3.440^{***}	3.779^{***}	3.440^{***}
	(0.0853)	(0.102)	(0.0852)	(0.102)	(0.0854)	(0.102)	(0.0853)	(0.102)
Stake	0.00449^{***}	0.00567^{***}	0.00463^{***}	0.00590^{***}	0.00454^{***}	0.00574^{***}	0.00454^{***}	0.00574^{***}
	(0.000782)	(0.00102)	(0.000788)	(0.00103)	(0.000781)	(0.00102)	(0.000784)	(0.00102)
Developed	0.536^{***}	0.560^{***}			0.505^{***}	0.505^{***}	0.496^{***}	0.500^{***}
	(0.107)	(0.129)			(0.0983)	(0.120)	(0.0981)	(0.120)
ExtraEU	0.0398	0.0669	-0.147^{**}	-0.144^{*}				
	(0.0628)	(0.0742)	(0.0647)	(0.0771)				
Mark.Cap./(%GDP)			0.00119^{**}	0.00135^{**}				
			(0.000479)	(0.000572)				
Common Law					-0.0963*	-0.0508		
					(0.0582)	(0.0693)		
Distance (capital)							-6.12e-06	-3.32e-06
							(8.98e-06)	(1.04e-05)
Observations	4,300	2,880	4,275	2,868	4,294	2,874	4,294	2,874
Constant	yes	yes	yes	yes	yes	yes	yes	yes
Year dumnies	yes	yes	yes	yes	yes	yes	yes	yes
Pseudo R2	0.320	0.276	0.319	0.275	0.320	0.276	0.320	0.276
F-test	59.92	52.77	41.34	40.10	59.66	50.70	58.22	50.44
$\operatorname{Prob}(\operatorname{F-test}) > \operatorname{F}$	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
NOTES: Dependent varia	able: NBF_V	$S_BANK.$ The	sample consis	sts of the FDI 1	recipient firms	used in the pa	nel regressions.	Mark.Cap.
stands for the average bet	tween 1998 and	l 2016 of the val	ue of stocks tra	aded in capital 1	markets over C	DP in the count	try of origin of	the investor.
Standard errors in parent	theses. *** p<	0.01, ** p < 0.05	, * p < 0.10					

Table 13: The impact of distance and financial sophistication on non-bank indebtedness. Extended panel. Fractional regressions. Dependent variable: NBF_VS_BANK

7 Additional results and robustness analysis

In this section we provide further evidence on the effects of the foreign investment on the financial profile of target firms. Moreover, to check the robustness of our results, we focus on two aspects of our analysis, namely the extent to which the entry of a foreign investor relaxes the target firm's financing constraint, and whether the effects on the investment process differ in the case of a domestic acquisition as opposed to a foreign one.

7.1 FDIs, financial structure and borrowing costs

In the previous sections we have shown that FDI recipient firms rotate their financial structure, favoring non-bank financial sources. This is reflected by a significant increase in the non-bank financing over total financing ratio that consolidates over time. Thanks to the increased orientation towards such alternative sources, among the other reasons, FDI targets show an extended investment capacity also with regards to intangible assets.

In this section we investigate whether, apart from the increased investment effect, the greater orientation towards non-bank financing entails also a reduction of borrowing costs for FDI target firms. In Table (14) we present results coming from fixed effects specifications that include time dummies. The dependent variable is the natural logarithm of financial expenditures, as reported in firms' income statements. Studies on borrowing costs usually employ loan spreads as a dependent variable (Lin et al., 2011; Ertugrul et al., 2017). However, since we have detailed information on balance sheet data, we recur to financial expenditures as defined by the IAS-23 item "Borrowing costs". Using such variable in an analysis on borrowing costs is not novel (see Zou and Adams, 2008; Luo et al., 2019). Moreover, it has the advantage to consider not only interest expenses, but also all different fees that financial intermediaries charge on borrowers, a relevant component of borrowing costs (Berg et al., 2016).

Results from Table (14) point to the positive role of foreign investors in reducing target firms' financial expenditures. In the first column we only include the FDI_after dummy and control variables as regressors. The arrival of the foreign investor is associated to a drop in financial expenditures of about 19 % on impact. We then include the ratio of non-bank financial debt over total financial debt (NBF_VS_BANK) as an additional covariate. Both the regressors of interest show a negative and significant coefficient, indicating that ceteris paribus foreign investors are able to reduce target firm's financial expenditures of 10 %, and that an increase of NBF_VS_BANK , for instance, from 30 to 31 per cent is associated to a reduction of financial expenditures of 0.4%.²⁹ Results

²⁹According to our estimates, hence, a unit increase in non-bank financing over total financial debt would be associated to a decrease of financial expenditures of 40%. This comes from the usual interpretation of log-linear models, where a unit increase of the independent variable is associated to an increase/reduction of the dependent of $100^{*}\beta\%$. Since $NBF_{-}VS_{-}BANK$ is a ratio that varies between 0 and 1 we prefer to interpret the result in terms of an increase of one percent, so that the coefficient associated to the variable must not be multiplied by 100 to interpret its impact on the dependent variable, i.e. a one percent increase of $NBF_{-}VS_{-}BANK$

Table 14: Panel regression results. Dependent variable: Borrowing costs (log)

	k=0	k=0	k=1	k=2	k=3
FDI_{t-k}	-0.189***	-0.100***	-0.160***	-0.151***	-0.133***
	(0.022)	(0.021)	(0.022)	(0.024)	(0.026)
$NBF_VS_BANK_{t-k}$		-0.389***	-0.394***	-0.323***	-0.327***
		(0.020)	(0.021)	(0.022)	(0.023)
$Cashflow_t$	-0.016^{***}	-0.015^{***}	-0.014^{***}	-0.015***	-0.015***
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
$Totalasset(log)_t$	0.806***	0.802***	0.765^{***}	0.749^{***}	0.732^{***}
,	(0.019)	(0.019)	(0.021)	(0.023)	(0.027)
$Totalasset(log)_{t-1}$	0.378^{***}	0.363^{***}	0.394^{***}	0.440^{***}	0.485^{***}
	(0.016)	(0.016)	(0.019)	(0.021)	(0.025)
$Valueadded(growth)_t$	$5.0e-04^{**}$	$5.0e-04^{**}$	9.0e-04***	-7.0e-04***	$3.0e-04^{***}$
	(2.0e-05)	(2.0e-05)	(2.0e-05)	(1.0e-04)	(2.0e-04)
Observations	25,984	23,457	23,260	21,502	20,033
Number of firms	1,832	1,826	1,821	1,802	1,787
Constant	yes	yes	yes	yes	yes
Year dummies	yes	yes	yes	yes	yes
R^2 within	0.338	0.340	0.337	0.327	0.321
R^2 overall	0.673	0.710	0.704	0.690	0.677
F-test	329.7	285.7	278.6	251.9	232.4
Prob(F-test) > F	0.000	0.000	0.000	0.000	0.000

NOTES: Period 1998-2016.Standard errors in parentheses.

*** p<0.01, ** p<0.05, * p<0.10

are stable when we impose different lag structures to the main regressors in order to capture the dynamics of such finding (column 3 to 5).

In Table (15) we dig deeper and estimate fixed effects specifications similar to equation (3). The dependent variable is again the natural log of financial expenditures, while the main regressors are the amount of non-bank financing and the amount of bank debt, both divided by the average total assets at firmlevel, and the *FDI_after* dummy. The aim of the models is to better understand the previous finding, i.e. the reduction of financial expenditures deriving from an increase of NBF_VS_BANK. Column 1 shows that FDI target firms witness a reduction of financial expenditures of about 13% with respect to control firms. Moreover, while an increase of one percent in the amount non-bank financing over the average total assets of the firm is associated to an increase of about 0.95% of financial expenses, bank debt is more expensive, producing an increase of financial expenses of about 1.46%. Results are even sharper when we include interactive terms (column 2 to 4). In this case, the FDI_after dummy is still associated to a negative and significant coefficient. In addition, the effect of non-bank financing is still positive and lower than that of bank debt, but it is attenuated further by the presence of the foreign investor. Conversely, it appears that when recurring to bank debt, that is already more expensive, FDI investors incur in additional costs. Again, such results point to a positive role played by foreign investors in reducing the cost of borrowing of target firms, in

produces an increase/reduction of borrowing costs of $\beta\%$

Table 15: Panel regression results. Dependent variable: Borrowing costs (log)

FDI	-0.133***	-0.116***	-0.173***	-0.155***
	(0.0189)	(0.0205)	(0.0241)	(0.0258)
Non-bank financing	0.954^{***}	1.025^{***}	0.954^{***}	1.019^{***}
	(0.0356)	(0.0489)	(0.0356)	(0.0489)
Bank debt (on asset)	1.462^{***}	1.462^{***}	1.419^{***}	1.421^{***}
	(0.0296)	(0.0296)	(0.0338)	(0.0338)
Bank debt (growth)	-3.58e-06	-3.11e-06	-3.48e-06	-3.06e-06
	(1.22e-05)	(1.22e-05)	(1.22e-05)	(1.22e-05)
Cash flow	-0.00827***	-0.00829***	-0.00824^{***}	-0.00826***
	(0.000664)	(0.000664)	(0.000664)	(0.000664)
Total asset (log)	0.224***	0.222***	0.226***	0.225***
	(0.0224)	(0.0224)	(0.0224)	(0.0225)
Total asset (log, lag)	0.520***	0.520***	0.520***	0.520***
	(0.0198)	(0.0198)	(0.0198)	(0.0198)
Value added (growth)	1.80e-08	2.10e-08	1.88e-08	2.14e-08
	(1.32e-07)	(1.32e-07)	(1.32e-07)	(1.32e-07)
FDI#Non-bank financing		-0.167**		-0.154**
		(0.0647)		(0.0649)
FDI#Bank debt (on asset)			0.139^{***}	0.131**
			(0.0523)	(0.0525)
Observations	20,850	20,850	20,850	20,850
Number of firms	1,775	1,775	1,775	1,775
Constant	yes	yes	yes	yes
Year dummies	yes	yes	yes	yes
R^2 within	0.436	0.436	0.436	0.436
R^2 overall	0.782	0.782	0.782	0.782
F-test	170.2	158.2	157.7	157.7
Prob(F-test) > F	0.000	0.000	0.000	0.000

NOTES: Period 1998-2016. Standard errors in parentheses. *** p0.01, ** p0.05, * p0.10

line with Wu et al. (2019), and the relevance of non-bank instruments in such context.

7.2 More on the target firm's financial constraint

In section 5 we have shown that target firms seem to be less financially constrained in the aftermath of a foreign acquisition by looking at how they finance their investments, with a particular attention to those that are easier to finance through bank borrowing, namely those in tangible assets. In that context, investment shows little sensitivity to cash flow when the firm is invested in from abroad. To gauge the extent of the firms' financing constraint unrelated to their investment decision, we follow Almeida et al. (2004) and Erel et al. (2015) and estimate the cash flow sensitivity of cash for FDI targets and control firms. The authors argue that it is possible to infer whether firms are financially constrained by analyzing their response to an incremental cash flow in terms of cash holding. In detail, while constrained firms react to supplementary cash flow by saving more cash to weather potential future storms, this pattern is not observed for unconstrained firms. The latter can indeed finance investment by accessing financial markets, so that they do not need to modify their saving schemes as a result of an increase in cash flow. What is relevant for the purpose of our study is "the change in the cash flow sensitivity of cash around the time of the acquisition" (Erel et al. (2015), p. 305), as this would reflect any effect of the foreign operation on the financial constraint faced by target firm. Conversely, we estimate different specifications based on the following equation:

$$\Delta CashHolding_{i,t} = \alpha + \beta_1 FDI_after_{i,t} + \beta_2 CashFlow_{i,t} + \beta_3 (FDI_after_{i,t} \times CashFlow_{i,t}) + \Gamma X_{i,t} + c_i + \epsilon_{i,t}$$
(5)

where $\Delta CashHolding_{i,t}$ is the annual change in (log) cash (and related assets) over average total assets. The equation includes firm-specific controls, year dummies, and firm-fixed effects. All the covariates entering the equation are scaled by average total assets to avoid feedback effects. Our parameters of interest here are β_2 and β_3 , representing the cash flow sensitivity of cash before the acquisition and the effect of the foreign investor on the latter, respectively. We expect a positive β_2 and a negative β_3 , implying that target firms are financially constrained before the acquisitions and that the foreign investor relaxes (and possibly offsets) the constraint. Results are presented in Table (16).

In column 1 we report our benchmark specification, including the FDI dummy, cash flow and the interaction between the two. Current and lagged total assets (in log), annual variation in value added, time dummies, and firm fixed effects complete the specification. We then progressively add from column 2 to column 4 non-bank financing, the annual percentage variation of bank debt, the lagged stock of bank debt, the dummy Recap and their interactions with FDI, all sources of funding directly linked to possible financing constraint. In column 5 we follow Almeida et al. (2004) and estimate a specification that considers the change of the firm's cash holding as a function of a number of sources and (competing) uses of liquid funds. The latter comprises the annual variation in working capital, the annual variation in short term financial debt (bank and non-bank debt) and current expenditures that in our case include both capital expenditures and acquisitions. Almeida et al. (2004) recognize that endogeneity may affect the last specification, hence they adopt an instrumental variable (IV) approach and instrument the variation of working capital and of short term debt, and the variables reflecting investment decisions. Similarly, in the last column we employ such a technique and instrument the additional variables with lagged fixed assets, lagged expenditures, lagged investment, lagged short term debt and lagged net income.

Results are not particularly sensitive to different specifications, suggesting that before the foreign acquisition target firms used to hold as cash a positive fraction of the incremental cash flow. This is captured by the positive and significant coefficient associated to cash flow. On the other hand, the arrival of the FDI provides fresh capital, a greater financial diversification and improved

	OLS	OLS	OLS	OLS	OLS	IV
FDI	0.0423^{***}	0.0570^{***}	0.0492^{***}	0.0350 (0.0224)	0.0394^{***}	0.122 (0.136)
Cash flow	(0.0113) 0.00520^{***} (0.000444)	(0.0125) 0.00514^{***} (0.000447)	(0.0102) 0.00568^{***} (0.000465)	(0.0224) 0.00570^{***} (0.000465)	(0.0013) 0.00423^{***} (0.000425)	(0.130) 0.0120^{*} (0.00647)
FDI#Cash flow	-0.00270^{***} (0.000658)	-0.00278*** (0.000659)	-0.00321^{***} (0.000674)	-0.00312^{***} (0.000675)	-0.00264^{***} (0.000610)	-0.00302^{*}
Non-bank financing	(0.0000000)	(0.00378) (0.0244)	-0.0182 (0.0277)	-0.0170 (0.0277)	(01000010)	(0100101)
${\rm FDI}\#{\rm Non\text{-}bank}$ financing		-0.0889*** (0.0339)	-0.0442 (0.0373)	-0.0442 (0.0373)		
Bank debt (growth)		(0.0000)	-4.18e-06 (1.38e-05)	-4.37e-06 (1.38e-05)		
FDI#Bank debt (growth)			7.20e-06 (1.60e-05)	7.29e-06 (1.60e-05)		
Bank debt (on asset,lag)			0.0626^{***} (0.0200)	0.0643^{***} (0.0201)		
FDI#Bank debt (on asset,lag)			0.00515 (0.0317)	0.00388 (0.0318)		
Recap			. ,	0.0300*** (0.0112)		
FDI#Recap				0.0155 (0.0214)		
Working capital (growth)				· /	0.415^{***} (0.0152)	-2.437 (2.141)
Short term debt (growth)					-0.801*** (0.0133)	0.501 (1.123)
Expenditures					-0.00633 (0.00848)	-0.506 (0.947)
Observations	26,938	26,938	21,130	21,130	23,042	19,276
Number of firms	1,839	1,839	1,781	1,781	1,836	1,803
Constant	yes	yes	yes	yes	yes	yes
Year dummies	yes	yes	yes	yes	yes	yes
Additional firm-level controls	yes	yes	yes	yes	yes	yes
R^2 within	0.018	0.019	0.016	0.017	0.182	na-
R^2 overall	0.002	0.002	0.002	0.002	0.079	0.022
F-test	12.10	11.80	7.089	6.985	120.4	142.72
Prob(F-test)>F	0.000	0.000	0.000	0.000	0.000	0.000

Table 16: Panel regression results. Dependent variable: $\Delta CASH_HOLDING$

NOTES: Period 1998-2016. Additional firm-level controls: current and lagged total assets (in log), annual variation of value added. Standard errors in parentheses. *** p0.01, ** p0.05, * p0.10

access to financial markets, as we have documented in the previous sections. This, in turn, results in a negative and significant coefficient for the interaction term β_3 in Table (16), i.e. the cash flow sensitivity of cash for target firms is reduced after the acquisition. All in all, the above findings corroborate the hypothesis that foreign investors relax the target firm's financial constraint after the operation.

7.3 What if the investor is domestic?

We have seen in the previous sections that domestic M&A operations do not show a relevant rotation in the firm's financial structure, in this part of the paper we will see what happens to investment. Regression results are reported in Table (17). Unlike in the context of the foreign acquisition, the increase in shareholders' capital is positively linked to total and intangible investments, but its effect is magnified if this happens in connection with a domestic acquisition. Apparently, a firm's recapitalization and cash flow are important sources used to finance investment prior to the arrival of a new investor in the ownership structure. As for the relevance of bank debt, results are ambiguous. However, in any case the occurrence of a domestic acquisition does not mitigate the role of such source. On the contrary, non-bank channels appear to be the main tool to finance investments; the arrival of the new investors partially offsets such reliance, in particular for financing of tangibles. This marks the main difference with the foreign acquisition, where regression results suggested that the new ownership may have brought new resources, new key people in the firm's management or both and this may have been reflected also in a shift in the financing of the firm and its investments in favor of alternative financing sources. In the case of domestic acquisition, however, this effect does not materialize extensively.

	Total	Tangible	Intangible
DOMESTIC	0.0152^{**}	0.00546	0.00969**
DOMESTIC	(0.00592)	(0.00451)	(0.00409)
Perep	0.00778**	0.00269	0.00508**
песар	(0.00321)	(0.00244)	(0.00222)
Cash from	0.000257^{**}	-0.000207**	0.000464***
Cash now	(0.000130)	(9.91e-05)	(9.01e-05)
Non-bonk financing	0.0166**	0.00997^{*}	0.00667
Non-Dank imancing	(0.00786)	(0.00598)	(0.00543)
Book date (month)	$2.58e-06^{*}$	2.06e-06**	5.19e-07
Dank debt (growth)	(1.33e-06)	(1.02e-06)	(9.22e-07)
Deal delt (an area las)	-0.0336***	-0.0351***	0.00158
Bank debt (on asset, lag)	(0.00567)	(0.00432)	(0.00392)
DOMESTIC // Bassar	0.0142^{**}	0.00381	0.0104***
DOMESTIC#Recap	(0.00579)	(0.00440)	(0.00400)
DOMECTIC // C L. A.	0.00107***	-0.000317**	0.00139***
DOMESTIC#Cash flow	(0.000190)	(0.000144)	(0.000131)
	0.00537	-0.0207**	0.0261***
DOMESTIC#Non-bank fin.	(0.0108)	(0.00823)	(0.00748)
	0.000199^{***}	$-1.38e-05^{**}$	0.000213***
DOMESTIC#Bank debt growth	(9.22e-06)	(7.01e-06)	(6.37e-06)
	-0.0474***	-0.0176***	-0.0298***
DOMESTIC#Bank debt (on asset, lag)	(0.00828)	(0.00629)	(0.00572)
	. /	. ,	× /
Observations	30,543	30,543	30,543
Number of firms	2,661	2,661	2,661
Constant	yes	yes	yes
Year dummies	yes	yes	yes
Additional firm-level controls	yes	yes	yes
R 2 within	0.190	0.109	0.133
R 2 overall	0.0480	0.0250	0.0440
F-test	135.7	70.65	89.11
Prob(F-test) > F	0.000	0.000	0.000

Table 17: Panel regression results. Domestic acquisitions

NOTES: Period 1998-2016. Additional firm-level controls: current and lagged total assets (in log), annual variation of value added. Standard errors in parentheses. *** p0.01, ** p0.05, * p0.10

8 Concluding remarks

In this paper, we have investigated the impact of foreign acquisition from a corporate finance point of view on a large sample of Italian firms. As foreign investors arrive, the firms' financial structure rotates in favor of non-bank financing even in a context, like the Italian one, of not fully developed capital markets. This result suggests that after a takeover from abroad, the target company shows a greater willingness to use non-traditional funding means. In our opinion, this fact may reflect a higher level of managerial sophistication stemming from being part of a larger, more internationalized group. This hypothesis gains traction from two facts. First, the time profile of the rotation effect, that is amplified over time, suggesting that a gradual process of managerial overhaul is at play after the foreign acquisition. Second, the same process occurs to a lower extent if the acquirer company is a domestic one, which probably shares with the target many features in the corporate culture, including a lower willingness to diversify the financing structure.

Is this relevant for investment? In our paper, we answer positively to this question, although our findings also point to a more nuanced figure. Indeed, the amount of non-bank financing relates positively to total investment and, to a lesser extent, to its "tangible" component. Furthermore, the investment process appears to benefit from a foreign acquisition due to the availability of new resources. This, in turn, reduces the sensitivity of total investment to bank debt and of tangible investment to cash flow (and to bank debt). Finally, we detect that also the accumulation of intangible asset may benefit from the arrival of the cavalry. The latter greatly benefits from the increased propensity to tap non-bank financing that comes with the presence of a foreign investor.

The impact of foreign direct investment is a widely debated issue. In this paper, we contribute to the comprehension of this process. Possible follow up may touch upon the managerial structure of the firms, to see if the results we found in our paper relate more to a greater availability of resources or are underpinned by a deeper restructuring of the acquired firm's managerial practices and culture. While in both cases foreign investors would be "good" for the economy on this specific ground, for a policy maker depending on which of the two channels is dominant a different policy answer may be preferable. While in the first case, the optimal response goes in the direction of encouraging firms to access capital markets, in the second the policymaker should push firms to go through a much deeper restructuring by adopting different and more sophisticated managerial standards.

References

Agarwal, S., and R. Hauswald. (2010). Distance and private information in lending. *The Review of Financial Studies*, 23, 2757–88.

Alessandrini, P., Presbitero, A. F., and Zazzaro, A. (2009). Banks, Distances and Firms' Financing Constraints. *Review of Finance*, 13(2), 261–307.

Alfaro, L., Chanda, A., Kalemli-Ozcan, S., and Sayek, S. (2004). FDI and economic growth: the role of local financial markets. *Journal of International Economics*, 64(1), 89-112.

Almeida, H. and Campello, M. (2007). Financial Constraints, Asset Tangibility, and Corporate Investment. *Review of Financial Studies*, 20 (5), 1429-1460.

Almeida, H., Campello, M. and Weisbach, M. S. (2004). The cash flow sensitivity of cash. *Journal of Finance*, 59, 1777–1804.

Alquist, R., Berman, N., Mukherjee, R. and Tesar, L. L. (2019). Financial constraints, institutions, and foreign ownership. *Journal of International Economics*, 118, 63-83.

Amiti, M., and Weinstein, D. E. (2018). How Much Do Idiosyncratic Bank Shocks Affect Investment? Evidence from Matched Bank-Firm Loan Data. *Journal of Political Economy*, 126, 525-587.

Anwar, S., and Sun, S. (2015). Can the presence of foreign investment affect the capital structure of domestic firms? *Journal of Corporate Finance*, 30, 32-43.

Barba Navaretti, G., & Castellani, D. (2004). Investments abroad and performance at home: Evidence from Italian multinationals. *(CEPR Discussion Papers 4284)*, Centre for Economic Policy Research.

Bentivogli, C., and Mirenda, L. (2017). Foreign ownership and performance: evidence from Italian firms. *International Journal of the Economics of Busi*ness, 24(3), 251-273.

Berg, T., Saunders, A., and Steffen, S. (2016). The total cost of corporate borrowing in the loan market: Don't ignore the fees. *The Journal of Finance*, 71(3), 1357-1392.

Bircan, Ç. (2019). Ownership structure and productivity of multinationals. *Journal of International Economics*, 116, 125-143.

Bollaert, H. and Delanghe, M. (2015). Securities Data Company and Zephyr, data sources for M&A research. *Journal of Corporate Finance*, 33, 85-100.

Bolton, P., Freixas, X., Gambacorta, L., and Mistrulli, P.E. (2016). Relationship and Transaction Lending in a Crisis. *The Review of Financial Studies*, 29(10), 2643-2676.

Borin, A. and Mancini, M. (2016). Foreign direct investment and firm performance: an empirical analysis of Italian firms. *Review of World Economics*, 152, 705-732.

Borisova, G. and Brown, J. R. (2013). R&D sensitivity to asset sale proceeds: New evidence on financing constraints and intangible investment. *Journal of Banking & Finance*, 37(1), 159-173.

Braguinsky, S., Ohyama, A., Okazaki, T., and Syverson, C. (2015). Acquisitions, productivity, and profitability: evidence from the Japanese cotton spinning industry. *American Economic Review*, 105(7), 2086-2119.

Brown, J. R., Fazzari, S. M. and Petersen, B. C. (2009). Financing Innovation and Growth: Cash Flow, External Equity, and the 1990s R&D Boom. *The Journal of Finance*, 64, 151-185.

Bugamelli, M., Lotti, F., Amici, M., Ciapanna, E., Colonna, F., D'Amuri, F., ... & Scoccianti, F. (2018). Productivity growth in Italy: a tale of a slow-motion change. *Bank of Italy QEF Occasional Paper*, n. 422.

Caliendo, M., & Kopeinig, S. (2008). Some practical guidance for the implementation of propensity score matching. *Journal of Economic Surveys*, 22(1), pp. 31–72.

Calligaris, S. (2015). Misallocation and TFP in Italy. Labour, 29, 367-393.

Cantillo, M. and Wright, J. (2000). How Do Firms Choose Their Lenders? An Empirical Investigation. *The Review of Financial Studies*, 13(1), 155–189.

Cecchetti, S. and Schoenholtz, K. (2018). Financing intangible capital. *CEPR Policy Portal VoxEU*.

CERVED, Cerved Group (2018), retrieved July, 2018 from https://www.cerved.com/en

Chen, Y., Hua, X. and Boateng, A. (2017). Effects of foreign acquisitions on financial constraints, productivity and investment in R&D of target firms in China. *International Business Review*, 26(4), 640-651. Cingano, F., Manaresi, F., & Sette, E. (2016). Does credit crunch investment down? New evidence on the real effects of the bank-lending channel. *The Review of Financial Studies*, 29(10), 2737-2773.

Dahlquist, M. and Robertsson, G. (2001). Direct foreign ownership, institutional investors, and firm characteristics. *Journal of Financial Economics*, 59, 413-440.

Dell'Ariccia, G., Kadyrzhanova, D., Minoiu, M. C., & Ratnovski, M. L. (2017). Bank lending in the knowledge economy. *International Monetary Fund Working Paper*, No. 234

Desai, M. A., Foley, C. F., and Hines Jr, J. R. (2004). A multinational perspective on capital structure choice and internal capital markets. *The Journal* of *Finance*, 59(6), 2451-2487.

Diamond, D. W. (1991). Monitoring and Reputation: The Choice between Bank Loans and Directly Placed Debt. *Journal of Political Economy*, 99(4), pp. 689-721.

Erel, I., Jang, Y. and Weisbach, M. S. (2015). Do Acquisitions Relieve Target Firms' Financial Constraints?. *Journal of Finance*, 70 (1), 289-328.

Ertugrul, M., Lei, J., Qiu, J., and Wan, C. (2017). Annual report readability, tone ambiguity, and the cost of borrowing. *Journal of Financial and Quantitative Analysis*, 52(2), 811-836.

Eun, C. S., Kolodny, R. and Scherega, C. (1996). Cross-border acquisitions and shareholder wealth: Tests of the synergy and internalization hypotheses. *Journal of Banking & Finance*, 20, pp. 1559-1582.

European Systemic Risk Board. (2014). Is Europed overbanked? Advisory Scientific Committee Report, 4.

Fatemi, A. M. (1988). The Effect of International Diversification on Corporate Financing Policy. *Journal of Business Research*, 16(1), 17-30.

Fazzari, S. M., Hubbard, G. R. and Petersen, B. C. (1988). Financing Constraints and Corporate Investment. *Brookings papers on economic activity*, 19 (1), 141-206.

Gertler, M. and Gilchrist, S. (1994). Monetary policy, business cycles, and the behavior of small manufacturing firms. *Quarterly Journal of Economics*, 109(2), 309-40.

Guadalupe, M., Kuzmina, O., and Thomas, C. (2012). Innovation and foreign ownership. *American Economic Review*, 102(7), 3594-3627.

Hale, G. and Santos, J. A. C. (2008). The decision to first enter the public bond market: The role of firm reputation, funding choices, and bank relationships. *Journal of Banking & Finance*, 32(9), 1928-1940.

Hardin, A., and Holmes, L. (2002). Measuring and modelling barriers to FDI. In Bijit Bota (ed.). *Foreign Direct Investment: Research Issues.* London, UK: Routledge.

Hattari, R., and Rajan, R. (2011). How Different are FDI and FPI Flows?: Distance and Capital Market Integration. *Journal of Economic Integration*, 26(3), 499-525.

Hauswald, R., and R. Marquez. (2006). Competition and strategic information acquisition in credit markets. *The Review of Financial Studies*, 19, 967–1000.

Huberman, G. (2001). Familiarity breeds investment. The Review of Financial Studies, 14(3), 659-680.

Javorcik, B. S. (2004). Does foreign direct investment increase the productivity of domestic firms? In search of spillovers through backward linkages. *American economic review*, 94(3), 605-627.

Kang, J.-K. and Stulz, R.M. (1997). Is Bank-Centered Corporate Governance Worth It? A Cross-Sectional Analysis of the Performance of Japanese Firms during the Asset Price Deflation. *NBER Working Paper*, 6238.

Kashyap, A K. and Stein, J. C. (2000). What Do a Million Observations on Banks Say about the Transmission of Monetary Policy? *The American Economic Review*, 90(3), 407-428.

Kashyap, A. K., Stein, J. C. and Lamont, O. (1994). Credit conditions and the cyclical behavior of inventories. *Quarterly Journal of Economics*, 109(3), 565-92.

Khatami, S. H., Marchica, M. T., and Mura, R. (2015). Corporate acquisitions and financial constraints. *International Review of Financial Analysis*, 40, 107-121.

Kiviet, J. F. (1995). On bias, inconsistency, and efficiency of various estimators in dynamic panel data models. *Journal of Econometrics*, 68(1), 53-78 La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. (1999). The Quality of Government. *Journal of Law, Economics and Organization*, 15(1):222-279.

Langfield, S. and Pagano, M. (2016). Bank bias in Europe: effects on systemic risk and growth. *Economic Policy*, 31(85), 51-106.

Lin, C., Ma, Y., Malatesta, P., and Xuan, Y. (2011). Ownership structure and the cost of corporate borrowing. *Journal of Financial Economics*, 100(1), 1-23.

Luo, Y. and Tung, R. L. (2007). International Expansion of EMERGING Market Enterprises: A Springboard Perspective. *Journal of International Business Studies*, 38(4), 481-498.

Luo, W., Guo, X., Zhong, S., and Wang, J. (2019). Environmental information disclosure quality, media attention and debt financing costs: Evidence from Chinese heavy polluting listed companies. *Journal of cleaner production*, 231, 268-277.

Mayer, T., & Zignago, S. (2011). Notes on CEPII's distances measures: The GeoDist database.

Naldi, L. and Davidsson, P. (2014). Entrepreneurial growth: The role of international knowledge acquisition as moderated by firm age. *Journal of Business Venturing*, 29(5), 687-703.

Orbis, Bureau Van Dijk (2018), retrieved July, 2018 from https://orbis.bvdinfo.com/version-201988/home.serv?product=OrbisNeo

Papke, L. E. and Woolridge, J. M. (1996). Econometric methods for fractional response variables with an application to 401 (k) plan participation rates. *Journal of Applied Econometrics*, 11, 619-632.

Rasciute, S. and Downward, P. (2017). Explaining variability in the investment location choices of MNEs: An exploration of country, industry and firm effects. *International Business Review*, 26(4), 605-613.

Rosenbaum, P., and Rubin, D. (1983). The central role of the propensity score in observational studies for causal effects. *Biometrika*, 70(1), 41-55.

Rutkowski, A. (2006) Inward FDI and Financial Constraints in Central and East European Countries. *Emerging Markets Finance and Trade*, 42(5), 28-60.

Slovin, M. B., Sushka, M.E. and Polonchek, J.A. (1993). The Value of Bank Durability: Borrowers as Bank Stakeholders. *The Journal of Finance*, 48(1),

247-266.

Stiebale, J., (2016). Cross-border M&As and innovative activity of acquiring and target firms. *Journal of International Economics*, 99, 1-15. http://dx.doi.org/10.1016/j.jinteco.2015.12.005

Stiebale, J. and Wößner, N. (2019). M&As, investment and financing constraints. *International Journal of the Economics of Business*, https://doi.org/10.1080/13571516.2019.1653719

Sun, Q. and Xiaolan, M. Z. (2019). Financing intangible capital. *Journal of Financial Economics*, https://doi.org/10.1016/j.jfineco.2019.04.003

UNCTAD (2018). Handbook of Statistics, Geneva 2018.

Veron, N. and Wolff, G. B. (2016). Capital Markets Union: A Vision for the Long Term. *Journal of Financial Regulation*, 2(1), 130-153.

World Bank (2018). World Development Indicators.

World Bank (2021). World Bank Country and Lending Groups. Retrieved from https://datahelpdesk.worldbank.org/knowledgebase/articles/906519

Wu, Y., Li, X., Zhang, Y., and Li, K. (2019). FDI inflow and financial costs of local firms. *Journal of the Asia Pacific Economy*, 24(4), 515-529.

Zephyr, Bureau Van Dijk (2018), retrieved July, 2018 from https://zephyr.bvdinfo.com/version-2019718/Home.serv?product=zephyrneo

Zou, H., and Adams, M. B. (2008). Debt capacity, cost of debt, and corporate insurance. *Journal of Financial and Quantitative Analysis*, 433-466.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1306 Effects of eligibility for central bank purchases on corporate bond spreads, by Taneli Mäkinen, Fan Li, Andrea Mercatanti and Andrea Silvestrini (November 2020).
- N. 1307 On the design of grant assignment rules, by Francesca Modena, Santiago Pereda Fernandez and Giulia Martina Tanzi (December 2020).
- N. 1308 Monetary policy strategies in the New Normal: a model-based analysis for the euro area, by Fabio Busetti, Stefano Neri, Alessandro Notarpietro and Massimiliano Pisani (December 2020).
- N. 1309 *Rare disasters, the natural interest rate and monetary policy*, by Alessandro Cantelmo (December 2020).
- N. 1310 The market stabilization role of central bank asset purchases: high-frequency evidence from the COVID-19 crisis, by Marco Bernardini and Annalisa De Nicola (December 2020).
- N.1311 *Equity tail risk in the treasury bond market*, by Dario Ruzzi and Mirco Rubin (December 2020).
- N. 1312 Scars of youth non-employment and labour market conditions, by Giulia Martina Tanzi (December 2020).
- N. 1313 *The COVID-19 shock and a fiscal-monetary policy mix in a monetary union*, by Anna Bartocci, Alessandro Notarpietro and Massimiliano Pisani (December 2020).
- N. 1314 Working horizon and labour supply: the effect of raising the full retirement age on middle-aged individuals, by Francesca Carta and Marta De Philippis (February 2021).
- N. 1315 Bank credit and market-based finance for corporations: the effects of minibond issuances, by Steven Ongena, Sara Pinoli, Paola Rossi and Alessandro Scopelliti (February 2021).
- N. 1316 Is inflation targeting a strategy past its sell-by date?, by Alberto Locarno and Alessandra Locarno (February 2021).
- N. 1317 Declining natural interest rate in the US: the pension system matters, by Jacopo Bonchi and Giacomo Caracciolo (February 2021).
- N. 1318 *Can we measure inflation expectations using Twitter?*, by Cristina Angelico, Juri Marcucci, Marcello Miccoli and Filippo Quarta (February 2021).
- N. 1319 Identifying deposits' outflows in real-time, by Edoardo Rainone (February 2021).
- N. 1320 Whatever it takes to save the planet? Central banks and unconventional green policy, by Alessandro Ferrari and Valerio Nispi Landi (February 2021).
- N. 1321 The power of text-based indicators in forecasting the Italian economic activity, by Valentina Aprigliano, Simone Emiliozzi, Gabriele Guaitoli, Andrea Luciani, Juri Marcucci and Libero Monteforte (March 2021).
- N. 1322 Judicial efficiency and bank credit to firms, by Giacomo Rodano (March 2021).
- N. 1323 Unconventional monetary policies and expectations on economic variables, by Alessio Anzuini and Luca Rossi (March 2021).
- N. 1324 *Modeling and forecasting macroeconomic dowside risk*, by Davide Delle Monache, Andrea De Polis and Ivan Petrella (March 2021).
- N. 1325 Foreclosures and house prices, by Michele Loberto (March 2021).
- N. 1326 *inancial structure and bank relationships of Italian multinational firms*, by Raffaello Bronzini, Alessio D'Ignazio and Davide Revelli (March 2021).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBANESE G., M. CIOFFI and P. TOMMASINO, *Legislators' behaviour and electoral rules: evidence from an Italian reform,* European Journal of Political Economy, v. 59, pp. 423-444, WP 1135 (September 2017).
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, v. 15, 4, pp. 55-80, WP 1098 (February 2017).
- ARNAUDO D., G. MICUCCI, M. RIGON and P. ROSSI, *Should I stay or should I go? Firms' mobility across banks in the aftermath of the financial crisis,* Italian Economic Journal / Rivista italiana degli economisti, v. 5, 1, pp. 17-37, **WP 1086 (October 2016).**
- BASSO G., F. D'AMURI and G. PERI, *Immigrants, labor market dynamics and adjustment to shocks in the euro area*, IMF Economic Review, v. 67, 3, pp. 528-572, WP 1195 (November 2018).
- BATINI N., G. MELINA and S. VILLA, *Fiscal buffers, private debt, and recession: the good, the bad and the ugly,* Journal of Macroeconomics, v. 62, **WP 1186 (July 2018).**
- BURLON L., A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of an open-ended asset purchase programme,* Journal of Policy Modeling, v. 41, 6, pp. 1144-1159, WP 1185 (July 2018).
- BUSETTI F. and M. CAIVANO, *Low frequency drivers of the real interest rate: empirical evidence for advanced economies*, International Finance, v. 22, 2, pp. 171-185, **WP 1132 (September 2017).**
- CAPPELLETTI G., G. GUAZZAROTTI and P. TOMMASINO, *Tax deferral and mutual fund inflows: evidence from a quasi-natural experiment*, Fiscal Studies, v. 40, 2, pp. 211-237, **WP 938 (November 2013).**
- CARDANI R., A. PACCAGNINI and S. VILLA, *Forecasting with instabilities: an application to DSGE models with financial frictions*, Journal of Macroeconomics, v. 61, WP 1234 (September 2019).
- CHIADES P., L. GRECO, V. MENGOTTO, L. MORETTI and P. VALBONESI, *Fiscal consolidation by intergovernmental transfers cuts? The unpleasant effect on expenditure arrears,* Economic Modelling, v. 77, pp. 266-275, WP 1076 (July 2016).
- CIANI E., F. DAVID and G. DE BLASIO, *Local responses to labor demand shocks: a re-assessment of the case of Italy*, Regional Science and Urban Economics, v. 75, pp. 1-21, WP 1112 (April 2017).
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, v. 8. 1, pp. 1-10, WP 985 (November 2014).
- CIAPANNA E. and M. TABOGA, *Bayesian analysis of coefficient instability in dynamic regressions*, Econometrics, MDPI, Open Access Journal, v. 7, 3, pp.1-32, WP 836 (November 2011).
- COLETTA M., R. DE BONIS and S. PIERMATTEI, *Household debt in OECD countries: the role of supply-side and demand-side factors,* Social Indicators Research, v. 143, 3, pp. 1185–1217, WP 989 (November 2014).
- COVA P., P. PAGANO and M. PISANI, *Domestic and international effects of the Eurosystem Expanded Asset Purchase Programme,* IMF Economic Review, v. 67, 2, pp. 315-348, WP 1036 (October 2015).
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, v. 23, 8. pp. 3457-2482, **WP 1174 (April 2018).**
- FERRERO G., M. GROSS and S. NERI, *On secular stagnation and low interest rates: demography matters,* International Finance, v. 22, 3, pp. 262-278, **WP 1137 (September 2017).**
- FOA G., L. GAMBACORTA, L. GUISO and P. E. MISTRULLI, *The supply side of household finance*, Review of Financial Studies, v.32, 10, pp. 3762-3798, **WP 1044 (November 2015).**
- GERALI A. and S. NERI, *Natural rates across the Atlantic,* Journal of Macroeconomics, v. 62, article 103019, WP 1140 (September 2017).
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, v. 78, pp. 118-133, WP 1167 (March 2018).
- GOMELLINI M., D. PELLEGRINO and F. GIFFONI, *Human capital and urban growth in Italy*, 1981-2001, Review of Urban & Regional Development Studies, v. 31, 2, pp. 77-101, WP 1127 (July 2017).
- LIBERATI D. and M. LOBERTO, *Taxation and housing markets with search frictions*, Journal of Housing Economics, v. 46, article 101632, WP 1105 (March 2017).
- MAGRI S., Are lenders using risk-based pricing in the Italian consumer loan market? The effect of the 2008 crisis, Journal of Credit Risk, v. 15, 1, pp. 27-65, WP 1164 (January 2018).
- MERCATANTI A., T. MAKINEN and A. SILVESTRINI, *The role of financial factors for european corporate investment,* Journal of International Money and Finance, v. 96, pp. 246-258, WP 1148 (October 2017).
- MIGLIETTA A., C. PICILLO and M. PIETRUNTI, *The impact of margin policies on the Italian repo market*, The North American Journal of Economics and Finance, v. 50, **WP 1028 (October 2015).**

- MONTEFORTE L. and V. RAPONI, Short-term forecasts of economic activity: are fortnightly factors useful?, Journal of Forecasting, v. 38, 3, pp. 207-221, WP 1177 (June 2018).
- NERI S. and A. NOTARPIETRO, Collateral constraints, the zero lower bound, and the debt-deflation mechanism, Economics Letters, v. 174, pp. 144-148, WP 1040 (November 2015).
- PANCRAZI R. and M. PIETRUNTI, *Natural expectations and home equity extraction*, Journal of Housing Economics, v. 46, 4, WP 984 (November 2014).
- PEREDA FERNANDEZ S., *Teachers and cheaters. Just an anagram?*, Journal of Human Capital, v. 13, 4, pp. 635-669, WP 1047 (January 2016).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, v. 23, 2, pp. 590-624, WP 871 (July 2012).

2020

- ALESSANDRI P. and M. BOTTERO, *Bank lending in uncertain times*, R European Economic Review, V. 128, WP 1109 (April 2017).
- ANTUNES A. and V. ERCOLANI, *Public debt expansions and the dynamics of the household borrowing constraint*, Review of Economic Dynamics, v. 37, pp. 1-32, WP 1268 (March 2020).
- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities*, Journal of Business & Economic Statistics, v. 38, 4, pp. 826-838, **WP 974 (October 2014).**
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, v. 16, 6, pp. 137-184, WP 1201 (November 2018).
- BOTTERO M., F. MEZZANOTTI and S. LENZU, *Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy,* Journal of International Economics, v. 126, article 103328, WP 1032 (October 2015).
- BRIPI F., D. LOSCHIAVO and D. REVELLI, Services trade and credit frictions: evidence with matched bank *firm data*, The World Economy, v. 43, 5, pp. 1216-1252, **WP 1110 (April 2017).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, Venture capitalists at work: a Diff-in-Diff approach at latestages of the screening process, Journal of Business Venturing, v. 35, 3, WP 1131 (September 2017).
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome,* Journal of Regional Science, v. 60, 4, pp. 801-822, WP 1208 (February 2019).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis*, Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, v. 102, **WP 1152 (December 2017).**
- D'ALESSIO G., *Measurement errors in survey data and the estimation of poverty and inequality indices,* Statistica Applicata - Italian Journal of Applied Statistics, v. 32, 3, WP 1116 (June 2017).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises,* Review of World Economic, v. 156, 4, pp. 859 885, WP 1217 (April 2019).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- ERCOLANI V. and F. NATOLI, *Forecasting US recessions: the role of economic uncertainty*, Economics Letters, v. 193, **WP 1299 (October 2020).**
- MAKINEN T., L. SARNO and G. ZINNA, *Risky bank guarantees*, Journal of Financial Economics, v. 136, 2, pp. 490-522, **WP 1232 (July 2019).**
- MODENA F., E. RETTORE and G. M. TANZI, *The effect of grants on university dropout rates: evidence from the Italian case,* Journal of Human Capital, v. 14, 3, pp. 343-370, WP 1193 (September 2018).
- NISPI LANDI V., *Capital controls spillovers*, Journal of International Money and Finance, v. 109, WP 1184 (July 2018).
- PERICOLI M., On risk factors of the stock-bond correlation, International Finance, v. 23, 3, pp. 392-416, WP 1198 (November 2018).
- RAINONE E., *The network nature of OTC interest rates*, Journal of Financial Markets, v.47, article 100525, WP 1022 (July 2015).

- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, WP 1059 (March 2016).
- RIZZICA L., *Raising aspirations and higher education. Evidence from the UK's widening participation policy,* Journal of Labor Economics, v. 38, 1, pp. 183-214, **WP 1188 (September 2018).**
- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, v. 24, 4, pp. 773-811, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, Bank Quality, judicial efficiency and loan repayment delays in Italy, Journal of Finance, v. 75, 4, pp. 2139-2178, WP 1072 (July 2016).

2021

- ALBANESE G., E. CIANI and G. DE BLASIO, *Anything new in town? The local effects of urban regeneration policies in Italy*, Regional Science and Urban Economics, v. 86, WP 1214 (April 2019).
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area*, Open Economies Review, v. 32, 1, pp. 71-107, **WP 1162 (January 2018).**
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, *A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds*, The Annals of Applied Statistics, v. 15, 1, pp. 304-322, **WP 1213 (March 2019).**
- LOSCHIAVO D., *Household debt and income inequality: evidence from Italian survey data*, Review of Income and Wealth. v. 67, 1, pp. 61-103, WP 1095 (January 2017).
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, v. 32, 1, pp. 183-211, WP 1200 (November 2018).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, v. 39, 2, pp. 575-588, WP 1092 (January 2017).

FORTHCOMING

- ACCETTURO A., A. LAMORGESE, S. MOCETTI and D. PELLEGRINO, *Housing Price elasticity and growth: evidence from Italian cities,* Journal of Economic Geography, WP 1267 (March 2020).
- ALBANESE G., G. DE BLASIO and A. LOCATELLI, *Does EU regional policy promote local TFP growth? Evidence from the Italian Mezzogiorno*, Papers in Regional Science, **WP 1253 (December 2019).**
- ANZUINI A. and L. ROSSI, Fiscal policy in the US: a new measure of uncertainty and its effects on the American economy, Empirical Economics, WP 1197 (November 2018).
- APRIGLIANO V. and D. LIBERATI, Using credit variables to date business cycle and to estimate the probabilities of recession in real time, The Manchester School, WP 1229 (July 2019).
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, Discretion and supplier selection in public procurement, Journal of Law, Economics, and Organization, WP 1178 (June 2018)
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, *Secular stagnation, R&D, public investment and monetary policy: a global-model perspective,* Macroeconomic Dynamics, **WP 1156 (December 2017).**
- DE PHILIPPIS M., *Multitask agents and incentives: the case of teaching and research for university professors,* Economic Journal, **WP 1042 (December 2015).**
- DEL PRETE S. and M. L. STEFANI, *Women as "Gold Dust": gender diversity in top boards and the performance of Italian banks*, Economic Notes, Monte dei Paschi di Siena, WP 1014 (June 2015).
- HERTWECK M., V. LEWIS and S. VILLA, *Going the extra mile: effort by workers and job-seekers,* Journal of Money, Credit and Banking, **WP 1277 (June 2020).**
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, **WP 1223 (June 2019).**
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility,* Journal of Human Resources, **WP 1182 (July 2018).**
- PERICOLI M. and M. TABOGA, Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models, Journal of Financial Econometrics, WP 1189 (September 2018).